MYTV Broadcasting Sdn. Bhd.

(Company No: 201001013318 / 897549-X)



Reference Access Offer

1 April 2024

Issued pursuant to the Malaysian Communications and Multimedia Commission Determination on Access List ('Access List'), Determination No. 6 of 2021, Commission Determination on the Mandatory Standard on Access ('MSA Determination'), Determination No. 1 of 2022 and Commission Determination on the Mandatory Standard On Access Pricing ('MSAP Determination'), Determination No. 1 of 2023

TABLE OF CONTENTS

INTRODUCTIO	N	3	
SECTION A: BA	ACKGROUND AND SCOPE OF THE RAO	4	
1.	INTRODUCTION	4	
2.	APPLICABILITY TO LICENSEES	4	
3.	NON-APPLICABILITY OF THE RAO	4	
SECTION B:	INTERPRETATION AND DEFINITIONS	5	
SECTION C:	PRINCIPLES OF ACCESS	8	
1.	LEGISLATIVE BACKGROUND	8	
2.	STANDARD ACCESS OBLIGATIONS		
SECTION D:	ACCESS REQUEST PROCESS AND PROCEDURES	9	
1.	INTRODUCTION	9	
2.	ACCESS REQUEST PROCESS	. 10	
3.	ACCEPTANCE	. 11	
4.	REQUEST FOR FURTHER INFORMATION	. 12	
5.	REJECTION OF ACCESS REQUEST	. 13	
6.	RIGHT TO REJECT	. 13	
7.	RIGHT TO WITHDRAW ACCESS REQUEST	. 14	
8.	APPLICABILITY FOR ADDITIONAL SERVICES	. 14	
9.	PROCESSING AND OTHER CHARGES	. 15	
10.	COMMENCEMENT OF NEGOTIATION	. 15	
11.	DURATION AND FORM OF NEGOTIATIONS	. 15	
12.	INITIAL MEETING	. 16	
SECTION E:	OBLIGATIONS FOR PROVISION OF INFORMATION	. 16	
SECTION F:	CONTENT OBLIGATIONS	. 17	
PART I -	FORECASTING OBLIGATIONS	. 18	
PART II - ORDERING AND PROVISIONING OBLIGATIONS21			
PART III - DECC	DMMISSIONING OBLIGATIONS	. 29	
PART IV - BILLI	NG AND SETTLEMENT OBLIGATIONS	. 30	
PART V - OPER	ATION AND MAINTENANCE OBLIGATIONS	. 33	
PART VI - TECH	INICAL OBLIGATIONS	. 36	
PART VII - TER	M, SUSPENSION AND TERMINATION OBLIGATIONS	. 37	
PART VIII - LEG	GAL BOILERPLATE OBLIGATIONS	. 40	
SECTION G:	SERVICE SPECIFIC OBLIGATIONS	. 41	
PART I -	DIGITAL TERRESTRIAL BROADCASTING MULTIPLEXING SERVICE (DTBMS) AND		
DIRECT-TO-HO	OME (DTH)	. 41	
PART II – NETV	NORK CO-LOCATION SERVICE	. 45	
PART III - CHARGES AND CHARGING PRINCIPLES			
	DISPUTE RESOLUTION PROCEDURES		
ANNEXURE 2:	CONFIDENTIALITY AND NON-DISCLOSURE AGREEMENT	. 61	
ANNEXURE 3:	APPLICATION FORM	. 66	
ANNEXURE 4: LIST OF TRANSMITTERS SITES70			

INTRODUCTION

- 1. This Reference Access Offer ("RAO") specifies the procedures and process to be followed by an Access Seeker who intends to acquire a Product from MYTV Broadcasting Sdn. Bhd. (Company No. 201001013318 / 897549-X) ("MYTV").
- 2. This RAO may be subject to amendments from time to time.
- 3. Where an amendment is made to the RAO, MYTV shall within twenty (20) Business Days supply an amended copy of the RAO to all Access Seekers who have submitted to MYTV an Access Request and which Access Request is still pending process by MYTV. The amendment to the RAO shall be effective after the expiry of the Initial Period if no notice of dispute is issued to and received by MYTV within the said period.
- 4. For the purposes of this RAO, an amendment shall mean an addition, deletion, or substitution to the provisions of the RAO other than an addition, deletion or substitution, which is undertaken to correct a typographical error or which is specifically mentioned in the RAO not to amount to an amendment of the RAO.
- 5. If the Commission revokes, varies or replaces the Access List and/or MSA Determination, and/or MSAP Determination in accordance with Section 56 of the Act, MYTV may, by giving written notice to all Access Seekers to whom it is supplying Access Service(s), withdraw or replace the RAO with effect from a date no earlier than the effective date of the Commission's revocation, variation or replacement.
- 6. In the event any amendment is made to the RAO, MYTV shall notify all Access Seekers of the effective date of the amendment.
- 7. All notices, forms, requests which are required to be sent by the provisions set out herein, shall be marked "RAO" and sent to the following:

MYTV Broadcasting Sdn. Bhd. 2271, Jalan Usahawan 2, Cyber 6, 63000, Cyberjaya, Selangor

Attention: Group Head Regulatory

A change of address shall not be construed as an amendment to the RAO.

8. The terminology used in this RAO has the meaning ascribed to them in **Section B**. All other words and phrases used in this RAO shall, unless the context otherwise requires, have the same meaning as in the MSA.

[The remainder of this page is intentionally left blank]

SECTION A: BACKGROUND AND SCOPE OF THE RAO

1. Introduction

1.1 This RAO only applies to Products/Services that are consistent with the terms of the Access List.

2. Applicability to Licensees

- 2.1 This RAO applies to Access Seekers who are licensed under the Act as:
 - (a) Content Applications Service Providers, and
 - (b) Applications Service Providers.
- 2.2 MYTV's RAO contains the terms and conditions for the Digital Terrestrial Broadcasting Multiplexing Services ("DTBMS"):
 - (a) Standard Definition Television Channel;
 - (b) High Definition Television Channel; and
 - (c) Digital Radio Channel.

3. Non-Applicability of the RAO

3.1 This RAO does not apply to Services which are not specified in the Access List.

[The remainder of this page is intentionally left blank]

SECTION B: INTERPRETATION AND DEFINITIONS

This **Section B** contains the meanings to words, phrases and expressions used in this RAO. Notwithstanding the foregoing, where a word or phrase or expression used in the RAO is given a specific meaning in or by the context of the RAO, such word, phrase or expression shall bear such meaning notwithstanding the contents of this **Section B**.

Access Agreement Means the bilateral agreement entered into between MYTV

and the Access Seeker which sets out the terms and conditions that govern the grant by MYTV of access to MYTV's

Services.

Access Provider means MYTV Broadcasting Sdn. Bhd. (Company No.

201001013318 / 897549-X)

Access List means the list of Facilities and Services determined from time

to time by the Commission pursuant to Chapter 3 of Part VI of the Act and the current Commission determination is Malaysian Communications and Multimedia Commission Determination on Access List, (Determination No. 6 of 2021) which came into operation on 15 December 2021, which sets out a list of Access Service(s) determined by the Commission

under Section 146 of the Act.

Access Seeker means an Operator who makes written request for access to

Services of MYTV or is being provided with Services by MYTV.

Act means the Communications and Multimedia Act 1998 (Act

588).

Applications Service bears the meaning ascribed to it under the Act.

Bank Guarantee means a guarantee executed in favour of MYTV, on behalf of

the Access Seeker, by a bank approved by MYTV and in a

format acceptable to MYTV.

Business Day means any day other than a Saturday, Sunday or public

holiday (whether gazetted or not) which is lawfully observed as a national public holiday and on which commercial banks are open for usual banking business in Kuala Lumpur and

Selangor.

Commission means the Malaysian Communications and Multimedia

Commission established under the Malaysian Communications and Multimedia Commission Act (Act 589) and also known as the Suruhanjaya Komunikasi dan

Multimedia Malaysia.

Content Applications Service bears the meaning ascribed to it under the Act.

Customer means in relation to a Party, a person having a contractual

relationship with the said Party for the provision of Applications Services including Content Applications Services

by means of that Party's Services.

DTBMS means Digital Terrestrial Broadcasting Multiplexing Service

referring to a Service for the combining of multiple content applications service Transport Streams into a single Transport Stream with or without the addition of Conditional Access information. DTBMS consists of compression, multiplexing, distribution and transmission processes which is integral of

the whole DTBMS system.

Direct-to-Home (DTH) a method of receiving digital TV service by means of signals

transmitted from broadcast satellites to personal dish

antenna located at the subscriber's premises.

Facilities means facilities which facilitates the provision of network

services or application services, including content application

services and "Facility" shall be construed accordingly.

Instrument means a direction or determination or declaration made by

the Minister or the Commission pursuant to the Act.

Licence means an Individual Licence or a Class Licence granted by the

Minister pursuant to the Act.

MSA Determination means the Malaysian Communications and Multimedia

Commission Determination On Mandatory Standard On

Access, Determination No. 1 of 2022.

MSAP Determination means the Malaysian Communications and Multimedia

Commission Determination on Mandatory Standard on Access

Pricing, Determination No. 1 of 2023

Network Facilities bears the meaning as ascribed in the Act.

Network Services bears the meaning as ascribed in the Act.

Operator means a network facilities provider, a network services

provider, an applications service provider or a content application service provider (as the context requires) and

includes both MYTV and the Access Seeker.

Party means MYTV or the Access Seeker as the context requires and

"Parties" means both MYTV and the Access Seeker.

Product means each of the separate provision by MYTV of access to its

Services and "Products" shall be construed accordingly.

RAO means the Reference Access Offer issued by MYTV.

Security Sum means security in the form of a Bank Guarantee provided or

to be provided by the Access Seeker to MYTV for the supply

of Access Service(s).

Services means the provision by MYTV of access to Services and

"Service" shall be construed accordingly.

SOC means Service Operation Centre of MYTV which monitors and

operate DTBMS system. SOC provide first level support for any defect on DTBMS that affect DTBMS availability and quality of service and manage support escalation process.

Third Party means a party who or which is not a party to the Access

Agreement.

[The remainder of this page is intentionally left blank]

SECTION C: PRINCIPLES OF ACCESS

1. Legislative Background

- 1.1 Pursuant to the issuance of the Malaysian Communications and Multimedia Commission Determination on Access List, (Determination No. 6 of 2021) ("Access List Determination"); and
- 1.2 Following the issuance of the Access List Determination, the Malaysian Communications and Multimedia Commission issued the Commission Determination on the Mandatory Standard on Access, Determination No. 1 of 2022 ("MSA Determination") which came into effect on 1st November 2022; and
- 1.3 Commission Determination on the Mandatory Standard on Access Pricing, Determination No. 1 of 2023 ("MSAP Determination") which came into effect starting from 1st March 2023, and
- 1.4 Pursuant to Section 5.3.3 of the MSA Determination, MYTV is pleased to prepare and maintain a Reference Access Offer ("RAO") in relation to Services on the Access List Determination which MYTV provides to itself or third parties and which:
 - (a) sets out the full terms and conditions on which MYTV is prepared to supply Services to Access Seekers, including the rates, charges, charging principles and methodologies to be applied for Services and any applicable fees or rebates;
 - (b) has description on Point of Interface;
 - (c) contain Copy of application form to be completed by the Access Seeker to apply for access to DTBMS;
 - (d) contains a copy of the MYTV's standard confidentiality and non-disclosure agreement; and
 - (e) contains only terms and conditions which are consistent with the rights and obligations set out in the MSA Determination and any applicable mandatory standard, including mandatory standard on Qos.

2. Standard Access Obligations

- 2.1 MYTV's RAO is consistent with:
 - (a) the standard access obligations stipulated under Section 4.1.1 of the MSA Determination and Section 149 of the Act; and
 - (b) the principles of reasonableness and non-discrimination stipulated under Sections 4.1.2, 4.1.3, 4.1.4, 4.1.5 and 4.1.6 of the MSA Determination.

SECTION D: ACCESS REQUEST PROCESS AND PROCEDURES

1. Introduction

- 1.1 An Access Seeker must comply with the following process if it wishes to obtain access to Services from MYTV.
- 1.2 Timing: If an Operator wished to negotiate an Access Agreement with MYTV:
 - (a) both parties shall notify the Commission when the negotiations for the Access Agreement begin under this paragraph;
 - (b) both parties shall use their best endeavours to conclude the Access Agreement within:
 - (i) where there is no Access Agreement in place between the operators, four (4) months; or
 - (ii) where there is already a commercial agreement or Access Agreemebt in place between the Operators, three (3) months,

after a written request by the Access Seeker to commence negotiations under Paragraph 5.4.6 (d) of the MSA and MYTV's response confirming it is willing to proceed to negotiate under Paragraph 5.4.7(b) of this RAO.

- (c) If the negotiations are not completed within the applicable timeframe specified under Paragraph 1.2(b) of this RAO:
 - (i) the parties may joinly apply to the Commission for an extension of time to negotiate and if the extension of time is not granted by the Commission, there shall be deemed to be a dispute between the parties and the Dispute Resolution Procedures shall take effect; or
 - (ii) either party may initiate the Dispute Resolution Procedures; and
- (d) If the Commission grants an extension of time under Paragraph 1.2 (c)(i) of this RAO, it may do so subject to such conditions as it specifies (such as an ongoing requirement to procide updates on negotiations at specified intervals and the right to reduce or extend any extension).
- 1.3 These provisions apply in the following circumstances:
 - (a) Where there is no current access agreement between MYTV and the Access Seeker; or
 - (b) If there is a valid and subsisting access agreement between MYTV and the Access Seeker and, either:
 - (i) such access agreement will expire within four (4) months from the date when the Access Seeker makes a request; or
 - (ii) the requested Service is outside the scope of such access agreement.

2. Access Request Process

- 2.1 An Access Seeker that wishes to obtain access to any Service from MYTV, must submit an Access Request to MYTV. The Access Request shall contain the following information;
 - (a) the name and contact details of the Access Seeker;
 - (b) the Services in respect of which access is sought;
 - (c) a list of relevant licences held by the Access Seeker;
 - (d) whether the Access Seeker wishes to accept the RAO, to negotiate amendments to the RAO or to negotiate an Access Agreement on alternative terms;
 - (e) the information (if any) the Access Seeker reasonably requires MYTV to provide for the purposes of the negotiations;
 - (f) two (2) copies of a confidentiality and non-disclosure agreement properly executed by the Access Seeker in the form prescribed by MYTV in Annexure 2;
 - (g) preliminary information regarding the scale and scope of Services that the Access Seeker expects to acquire from MYTV pursuant to the Access Request;
 - (h) relevant technical information relating to the interface Standards of the Access Seeker;
 - (i) creditworthiness information in accordance with MYTV's requirements;
 - (j) assessed security (or, if applicable, confirmation of security provided) in accordance with MYTV's requirements;
 - (k) insurance information in accordance with MYTV's requirement;
 - (I) bank guarantee information for security sum (12 months of service fee per channel); and
 - (m) such other information as MYTV may reasonably request for the sole purpose of providing access to the requested Services.
- 2.2 Within ten (10) Business Days of receipt of the Access Request from the Access Seekers, MYTV shall respond in writing to the Access Seeker by either:
 - (a) accepting the Access Request based on the terms and conditions in this RAO; or
 - (b) accepting the Access Request and to negotiate the amendments to the RAO or the Access Agreement on alternative terms; or
 - (c) requesting for further information from the Access Seeker; or
 - (d) rejecting the Access Request.

- 2.3 Information disclosure: In the event of Paragraph 2.2 (a) and (b) above, MYTV must provide the following information to an Access Seeker within ten (10) Business Days of receipt of a written request from that Access Seeker for the provision of access (whether or not on the basis of a RAO):
 - (a) any supplementary details of a Service offered by MYTV not included in the RAO, including details concerning all POIs and other locations (including sites deemed to be critical national information infrastructure and other secure sites) at which physical colocation, virtual co-location or in-span interconnection is available to Access Seekers;
 - (b) any supplementary access charges for access to Services not included in the RAO (for example, discounts for inferior service levels or surcharges for enhanced service levels);
 - (c) all supplementary technical information relating to the Services which may be the subject of the Access Request, which are not included in the RAO, including but not limited to any proof of concept (POC) information where available, physical and logical interfaces of its Network necessary to allow the development and deployment of communications services, value-added services and communications equipment that can interconnect to, and interoperate with, the MYTV's Network;
 - (d) supplementary details of the MYTV's operational processes and procedures not included in the RAO (e.g. regarding escorted access at sites deemed to be critical national information infrastructure or other secure sites);
 - supplementary details of the MYTV's provisioning cycles not included in the RAO and any impact such cycles may have upon an Access Request by the Access Seeker (e.g. capacity constraints);
 - (f) details of the MYTV's alternative quality of service targets not included in the RAO and actual achievements of service targets in respect of the Services which may be the subject of the Access Request;
 - (g) any security requirements, insurance requirements and creditworthiness information (including a credit assessment form, if available) required by MYTV under Sections 5.3.9, 5.3.10 and 5.3.11 of the MSA Determination; and
 - (h) MYTV's reasons for failing to supply any of the information referred to in Paragraphs 2.3(a) to 2.3.(g) of of this RAO.

Prior to the provision of information under Section 5.3.7 of the MSA Determination, MYTV may request the Access Seeker to enter into a confidentiality and non-disclosure agreement in accordance with Section 5.3.8 of the MSA Determination.

3. Acceptance

3.1 If the Access Request made by the Access Seeker accepts the terms and conditions of this RAO as the basis for the access to the Services offered to be provided by MYTV, then MYTV shall issue two (2) copies of the RAO to the Access Seeker which the Access Seeker shall execute

and return to MYTV within ten (10) Business Days together with one (1) copy of the executed confidentiality and non-disclosure agreement.

3.2 If:

- (a) the Access Seeker had submitted an Access Request which contains a request to negotiate the terms and conditions of the access agreement, and
- (b) MYTV is willing to proceed to negotiate the Access Request and the terms and conditions of the access agreement;

then MYTV shall do the following:

- execute the confidentiality and non-disclosure agreement and forward to the Access Seeker for signature and a copy of the executed confidentiality and nondisclosure agreement shall be forwarded to the Access Seeker;
- (ii) issue the draft Access Agreement; and
- (iii) specify a place, date and time no later than fifteen (15) Business Days from the date of MYTV's response, of which the Parties shall negotiate the terms and conditions of the Access Agreement; and
- (iv) require the Access Seeker to provide such additional information as may be reasonable in the circumstances.

4. Request for Further Information

- 4.1 MYTV may request the Access Seeker to provide further information on the Access Request.
- 4.2 The Access Seeker shall within ten (10) Business Days provide further information to MYTV as requested pursuant to Paragraph 4.1 above.
- 4.3 If in MYTV's opinion the information provided is sufficient for it to make a decision as to whether to accept or reject the Access Request, it shall reconsider the Access Request and inform the Access Seeker of its decision within ten (10) Business Days.
- 4.4 For the avoidance of doubt, MYTV may make more than one request for additional information from the Access Seeker in order for MYTV to make a decision as to whether to accept or reject the Access Request. Each request shall be subject to the provisions in Paragraph 4.2 and 4.3 above.
- 4.5 If the Access Seeker does not provide further information in response to a request made by MYTV, then the Access Seeker shall be deemed to have revoked the Access Request.

5. Rejection of Access Request

- 5.1 If MYTV rejects an Access Request, MYTV shall inform the Access Seeker and indicating the date its representatives are available to meet the Access Seeker. The Access Seeker may attend and meet with the representatives of MYTV on the date and time and at the venue specified which date shall not be later than seven (7) Business Days from the date of rejection notice of the Access Request.
- 5.2 If the Access Seeker fails or neglects to attend the specified meeting without providing acceptable reasons in writing at least one (1) day prior to the date of the meeting, then such failure shall be deemed to be an acceptance of the rejection of the Access Request.
- 5.3 If the Access Seeker does attend, and the rejection of the Access Request is not resolved (whether at that meeting or any subsequently agreed meeting), then either MYTV or the Access Seeker may initiate the Dispute Resolution Procedures set out in Annexure 1 of this RAO.
- 5.4 Pending the final determination of the dispute, MYTV shall not be obliged to provide access to the Access Seeker.

6. Right to Reject

- 6.1 MYTV may reject an Access Request made by an Access Seeker upon any of the following grounds:
 - (a) the information provided by the Access Seeker is incomplete or false; or
 - (b) it is not technically feasible to provide access to the Services requested; or
 - (c) MYTV has insufficient capacity or space to provide the requested Services; or
 - (d) MYTV reasonably believes that the Access Seeker may fail to make timely payments for the requested Services; or
 - (e) MYTV reasonably believes that the Access Seeker may fail, to a material extent, to comply with the terms and conditions of access of the relevant Services; or
 - (f) does not currently supply or provide access to the requested Services to itself or to any third party; or
 - (g) on the basis of national interest.
- 6.2 MYTV shall not refuse an Access Request on the grounds of unless MYTV establishes that there are substantial technical or operational concerns preventing the fulfilment of the Access Request. Each of the following matters shall be taken into account in determining whether access is technically feasible:

- (a) economic, accounting, billing, space or site concerns shall be disregarded by MYTV
 except that space or site concerns may be taken into account in circumstances where
 there is no possibility of expanding the space available on the relevant site;
- (b) any requirement for MYTV to modify its facilities or Equipment in order to meet the Access Request will not, on its own, mean that the access is not technically feasible;
- (c) if MYTV asserts that meeting the Access Request would have an adverse impact on network reliability, MYTV must provide evidence that provision of the requested Services would result in a specific and significant adverse impact on network reliability; and
- (d) MYTV must be able to demonstrate that it has considered and found not to be technically feasible (in accordance with this Paragraph) improvements that would allow MYTV to meet the Access Request (in whole, or in part, and including for an interim period until any primary difficulties can be resolved).
- 6.3 If MYTV rejects the Access Request, MYTV shall within ten (10) Business Days of receiving the Access Request:
 - (a) notify the Access Seeker in writing of MYTV's rejection;
 - (b) provide reasons for rejection under Paragraph 6.1 above to the Access Seeker;
 - (c) provide the basis for MYTV's rejection of the Access Request; and
 - (d) indicate a date and time, not later seven (7) Business Days from the date of this notice of rejection, at which representatives of MYTV will be available to meet with representatives of the Access Seeker to discuss the rejection of the Access Request.

7. Right to Withdraw Access Request

- 7.1 No later than the 5th Business Day following the date of the Access Request, the Access Seeker may withdraw the Access Request at no cost. The Access Seeker must inform MYTV in writing of its decision before the expiry of the 5th Business Day.
- 7.2 MYTV shall not be obliged to nor under any liability to fulfil an Access Request that is withdrawn.

8. Applicability for Additional Services

8.1 Notwithstanding that the Access Seeker may have entered into an Access Agreement with MYTV, the Access Seeker shall for each additional service of a type which it requires access to in addition or in substitution to the access already provided under an Access Agreement, submit an Access Request in accordance with the provisions set out above.

9. Processing and Other Charges

- 9.1 MYTV may charge the Access Seeker non-refundable fees for undertaking works to process the Access Request such as:
 - (a) Processing fee;
 - (b) Activation fee; and
 - (c) Additional DTBMS fee.
- 9.2 In the event additional and/or non-routine work i.e. non-administrative work or site visits or site surveys or technical surveys or technical evaluation work or testing etc, is required in order to process the Access Request, MYTV will charge a separate fee for undertaking such additional work.
- 9.3 The Access Seeker shall always remain liable to pay the fees for processing the Access Request and the undertaking by MYTV of the additional and/or non-routine work, notwithstanding the Access Request may have been rejected by MYTV or withdrawn by Access Seeker.

10. Commencement of Negotiation

- 10.1 If an Access Seeker has received a notice from MYTV to proceed with negotiation of the Access Agreement, the Access Seeker must within ten (10) Business Days of receipt of such notice submit to MYTV a list of its comments identifying the affected paragraphs with the proposed amendments, the priority of importance and the rationale for such change. The Access Seeker shall also specify in such submission, the date when it proposes to commence negotiations.
- 10.2 If the Access Seeker shall have failed, neglected and/or refused to do so within the stipulated period in Paragraph 10.1 above, then the Access Seeker shall have been deemed to have withdrawn its Access Request.

11. Duration and Form of Negotiations

- 11.1 All negotiations shall be concluded within one hundred and twenty (120) days from the date MYTV receives a written request to commence negotiations.
- 11.2 If negotiations are not completed within one hundred and twenty (120) days:
 - (a) the Parties may jointly apply to the Commission for an extension of time to negotiate and if the extension of time is not granted by the Commission, there shall be deemed to be a dispute between the Parties and the dispute resolution procedures under Annexure 1 shall take effect; or
 - (b) either Party may initiate the dispute resolution procedures.

11.3 Any meeting or negotiation may take place in person, or virtually by conference call, video conference or using other communications technology with participants in one or more geographical places (or in a combined form).

12. Initial Meeting

- 12.1 The designated representatives of MYTV and Access Seeker shall meet on the date and time at the venue specified by MYTV, and shall:
 - (a) agree a timetable for the negotiations, including milestones and dates for subsequent meetings;
 - (b) agree the negotiating procedures, including:
 - (i) the calling and chairing meetings;
 - (ii) the party responsible for keeping minutes of meetings;
 - (iii) clearly defined pathways and timetables for escalation within each party of matters not agreed in meetings;
 - (iv) procedures for consulting and including in the negotiating process relevant experts from the staff of each of the parties; and
 - (v) procedures for preparing and exchanging position papers;
 - (c) review the information requested and provided to date and identify information yet to be provided by each Party; and
 - (d) identify what technical investigations, if any, need to be made and by whom such investigations should be made.

SECTION E: OBLIGATIONS FOR PROVISION OF INFORMATION

- 1.1 The obligations of each Party to provide information to the other Party are subject to MSA Determination and the requirements of confidentiality in the confidentiality and non-disclosure agreement signed by the Parties.
- 1.2 An Access Seeker must provide MYTV on a timely basis with all agreed information reasonably required to determine the rates and charges to be billed by MYTV to the Access Seeker or by each Party to its Customers.
- 1.3 To the extent permitted by Malaysian laws and any relevant guidelines or Customer service standards in force pursuant to the Operator's respective Licence conditions, the Operator will exchange information and otherwise cooperate in relation to the prevention and investigation of fraudulent use or misuse of the Operator's respective communications services and/or theft of the Operator's provided terminal equipment.

- 1.4 Information provided under the RAO may only be used for the purpose for which it was given. Personal information about a Customer's credit worthiness, credit standing, credit history or credit capacity with any financial institution shall be treated with confidence and protected in accordance with Malaysian Laws.
- 1.5 Information required to be provided under the RAO need not be provided if the recipient Party has not established security measures that are adequate to protect the confidentiality of the information. If the recipient Party does not observe such security measures or any of the information is used by it for any purpose other than purpose for which it was given, the providing Party may deny the recipient Party further access to the information for the period during which the non-observance or non-conforming use of information is continuing. The Parties will cooperate to resolve the providing Party's reasonable concerns so that information exchange can be resumed as soon as possible.
- 1.6 The Parties acknowledge that when information (including for the purposes of this Paragraph any updated information) required to be provided under this paragraph is stored in a database, the Party entitled to receive the information will not be entitled to obtain direct access to the database. The precise method by which the information is to be made available will be determined by the Parties, having regard to the reasonable cost, convenience and security concerns of the Parties.
- 1.7 (a) Subject to the Act and any subordinate legislation, nothing in the Access Agreement may be construed as requiring a Party at any time to disclose to the other Party information which is at the date when the Access Agreement comes into force, the subject of a confidentiality obligation owed to the third person unless the third person consents to such disclosure. Where the consent of a third person is required, the Party holding the information must use its reasonable endeavours to obtain the consent of that third person.
 - (b) After the Access Agreement comes into force a Party must use its best endeavours not to enter into any contract which would prevent it from making relevant information available to the other Party unless the contract includes a term which permits the contracting Party to make the information available if directed to do so by the Commission.

SECTION F: CONTENT OBLIGATIONS

- 1.1 The Content Obligations referred to in this RAO are as follows:
 - (a) Forecasting Obligations as described in Part I of this Section;
 - (b) Ordering and Provisioning Obligations as described in Part II of this Section;
 - (c) Point of Interface Procedures as described in Part III of this Section;
 - (d) Decommissioning Obligations as described in Part IV of this Section;
 - (e) Network Change Obligations as described in Part V of this Section;
 - (f) Billing and Settlement Obligations as described in Part VI of this Section;

- (g) Operation and Maintenance Obligations as described in Part VII of this Section;
- (h) Technical Obligations as described in Part VIII of this Section;
- (i) Term, Suspension and Termination Obligations as described in Part IX of this Section;
- (j) Legal Boilerplate Obligations as described in Part X of this Section.

PART I - FORECASTING OBLIGATIONS

- 1. General
- 1.1 Part I of Section F sets out the forecasting procedures that are applicable only in relation to the provision of Services listed in the MYTV RAO preface.
- 2. Forecasting Requirements
- 2.1 As a result of special network management requirements, the Access Seeker is required to provide an eighteen (18) months' rolling forecast.
- 2.2 The minimum intervals or units of time to be used in Forecasts regarding DTBMS is six (6) months.
- 2.3 The maximum frequency to update or to make further Forecasts regarding DTBMS is once every six (6) months.
- 2.4 The network area or operational area to which Forecasts shall relate, which area shall correspond to that which MYTV uses for its own network planning and provisioning.
- 2.5 Such other information that MYTV reasonably requires in order to provide access to Services requested by the Access Seeker (which shall not include any information that MYTV does not provide to itself in connection with forecasting for its own services).
- 2.6 The Access Seeker shall meet the requirements of forecasting process that enables MYTV to plan for the expected needs of the Access Service(s) in order to fulfil the forecast.
- 2.7 Prerequisite information: The Access Seeker may request preliminary information from the MYTV about the availability and capacity of its Services to the extent the Access Seeker requires such information to provide Forecasts.
- 2.8 Confirmation of Forecast: If MYTV, acting reasonably will incur significant costs to ensure that access can be provided in accordance with a Forecast (for example, because it will need to proactively augment its Network to provide access within the requested timeframes), MYTV may request the Access Seeker to confirm the relevant Forecast. Once confirmed, the Forecast is deemed to be an Order pursuant to Part II of Section F of this RAO.
- 2.9 Non-binding: Subject to Paragraph 2.8 above, MYTV shall not require an Access Seeker to provide Forecasts that are legally binding on the Access Seeker, except to the extent that MYTV is permitted to recover costs and expenses as set out in Paragraph 2.18 below.

- 2.10 Forecast provision: MYTV may only require an Access Seeker to provide Forecasts in accordance with a Forecast Request no sooner than four (4) weeks after receipt of a Forecast Request.
- 2.11 Use of Forecast Information: Forecast Information provided by the Access Seeker shall be treated by MYTV as Confidential Information of the Access Seeker and shall only be used by those personnel of MYTV whose role is within either:
 - (a) MYTV's wholesale or interconnection group; or
 - (b) that part of the network engineering group of MYTV responsible for interconnection or access,

for the purpose of responding to and planning for the Forecast and related Orders. MYTV must maintain records that indicate which persons are provided with access to Forecast Information and, on request from the Commission, provide a copy of such records certified by the MYTV's Chief Operating Officer.

- 2.12 Distribution of Forecast Information: MYTV may only distribute Forecast Information of an Access Seeker outside the groups of people referred to in Paragraph 2.11 above if:
 - (a) the Forecast Information of the Access Seeker is aggregated with Forecasts provided by other Operators and the MYTV's own requirements (so as to protect the confidentiality of the Forecast Information); and
 - (b) the Forecast Information or its use does not otherwise identify the Access Seeker, its services or its Customers in any manner.
- 2.13 Time for response: MYTV must notify the Access Seeker within five (5) Business Days of receiving a Forecast whether or not MYTV considers the Forecast to be in compliance with the Forecast Request and:
 - (a) if, MYTV considers that the Forecast does comply with the Forecast Request, to specify in that notice the additional information which the Access Seeker is to provide to comply with the Forecast Request and MYTV will not require such information to be provided sooner than four (4) weeks after such a notice; or
 - (b) if, MYTV considers that the Forecast does comply with the Forecast Request, to specify in that notice that the Forecast is provisionally accepted subject to verification of the details of the Forecast and the matters set out in Paragraphs 2.14(a) to 2.14(d) below.
- 2.14 Reasons for rejection: MYTV may only reject a Forecast following provisional acceptance where MYTV reasonably believes that the Forecast is inaccurate or, there is insufficient capacity having regard to:
 - (a) the total current usage of the Services by MYTV and all Access Seekers;
 - (b) the current rate of growth of the Access Seeker's usage of the Services;

- (c) the current rate of growth of total usage of the Services by MYTV and All Access Seekers; and
- (d) subject to Paragraphs 2.24 and 2.25 of Part II Section F of this RAO, the amount of capacity in the Services that MYTV currently has available and can reasonably provision for the Access Seeker over the Forecast period, which must be at least equivalent to that which MYTV can reasonably provision for itself.
- 2.15 Time for acceptance or rejection: MYTV must give notice of any acceptance or rejection ("Rejection Notice") of a Forecast to the Access Seeker:
 - (a) within fifteen (15) Business Days of receipt of the relevant Forecast; and
 - (b) such Rejection Notice (if any) must specify:
 - (i) the grounds on which MYTV rejects the Forecast in accordance with Paragraph 2.14 above, at a sufficient level of detail to enable the Access Seeker to understand the basis of the rejection and to undertake its own re-assessment of the Forecast; and
 - (ii) an offer to meet within five (5) Business Days of the Rejection Notice of the Forecast to discuss the reasons for rejection and alternative methods of compliance. The meeting shall take place between MYTV and Access Seeker if the offer is accepted by the Access Seeker.
- 2.16 Reconsideration by Access Seeker: MYTV must allow an Access Seeker to reconsider its Forecast following a Rejection Notice and allow the Access Seeker, within twenty-one (21) Business Days of receipt of a Rejection Notice, either:
 - (a) to confirm its rejected Forecast, and explain why the Access Seeker considers that MYTV is obliged to accept the Forecast under this RAO; or
 - (b) to submit a new Forecast which the Access Seeker regards as meeting MYTV's concerns.
- 2.17 Reconsideration by Access Provider: MYTV shall reconsider any re-submitted or amended Forecast provided pursuant to Paragraph 2.16 and Paragraphs 2.13 to 2.15 above shall reapply.
- 2.18 Recovery for over-forecasting: MYTV shall not seek to recover any costs or expenses incurred due to its acceptance of a Forecast from an Access Seeker if the Forecast is not met by the Access Seeker unless:
 - (a) such costs and expenses were reasonably and necessarily incurred by MYTV;
 - (b) MYTV reasonably seeks to mitigate its loss (including through its own usage) provided MYTV shall not be required to do so for any greater period than the relevant Forecast period; and
 - (c) MYTV only recovers from the Access Seeker, seventy-five percent (75%) of such costs and expenses which could not be mitigated under Paragraph 2.18(b) above.

2.19 Meeting Forecasts: Subject to Paragraphs 2.13 to 2.15 above, MYTV must carry out network planning in order to enable Forecasts to be met. If an Access Seeker has confirmed a Forecast under Paragraph 2.8 above, it will be binding on the Access Seeker.

PART II - ORDERING AND PROVISIONING OBLIGATIONS

- 1. General
- 1.1 Part II of Section F sets out the ordering and provisioning procedures that are applicable only in relation to the provision of Services listed in the MYTV RAO preface.
- 2. Ordering Procedures
- 2.1 Contact point or mechanism: MYTV shall designate and notify an Access Seeker of one or more of the following:
 - (a) a person to whom Orders for access to Services are to be delivered;
 - (b) a contact point to which Orders for access to Services are to be delivered (such as an email address); and
 - (c) a mechanism where Orders for access to Services can be made (such as a web portal or B2B gateway), provided that if such a mechanism is the only method which MYTV provides for the receipt of Orders for that Service, MYTV cannot require the Access Seeker to unreasonably invest in specialised technology or systems (such as an automated interface between the Operational Support Systems ("OSS") of the Operators).
- 2.2 Subject to Paragraph 2.3 below the Access Seeker may place firm orders for DTBMS services from time to time with the quantity indicated in the first year forecast.
- 2.3 Order content: The Access Seeker shall ensure that the order contains the following information to enable MYTV to access and fulfil the order;
 - (a) the Services to which access is requested;
 - (b) a requested date and time for delivery;
 - (c) the location of the points of delivery;
 - (d) equipment of the Access Seeker to be used in connection with the Order, to the extent it may adversely affect the MYTV's Network; and
 - (e) such other information that MYTV reasonably requires in order for it to provision access to the Services as requested by the Access Seeker, provided that such information shall not include any information which:
 - (i) MYTV does not require from itself for similar provisioning;

- (ii) identifies, or which enables the identification of, a Customer or services of the Access Seeker; or
- (iii) is non-permitted information under Section 5.4.16 of the MSA.
- 2.4 Use of ordering information: Ordering information provided by Access Seeker shall be treated by MYTV as Confidential Information of the Access Seeker and shall only be used by those persons within MYTV whose role is within:
 - (a) MYTV's wholesale or interconnection group; and
 - (b) That part of the network engineering group of MYTV responsible for interconnection or access,

for the purpose of responding to and provisioning for the Order.

- 2.5 Treatment of Orders: MYTV shall:
 - (a) establish a single queue for all Orders for a given type of Service, whether those Orders are required for itself or any Access Seekers;
 - (b) give the equivalent priority to the handling of all Orders in each queue; and
 - (c) otherwise treat all Orders in each queue in compliance with its queuing policy established under Paragraph 2.22 below. .
- 2.6 Acknowledgement of receipt: MYTV shall acknowledge receipt of an Order for Services, in writing (or any other material or electronic form as agreed by the parties), within two (2) Business Days as specified in the Service Specific Obligations for the purposes of Paragraph 2.5.
- 2.7 Notice of Receipt: MYTV must include in its Notice of Receipt the following information:
 - (a) the time and date of receipt of the Order
 - (b) a list of any additional information reasonably required by MYTV from the Access Seeker to provision the Order;
 - (c) the position of the Order in MYTV's queue.
- 2.8 Acceptance Obligation: MYTV must use its reasonable efforts to accept and fulfil Orders from the Access Seeker for the Service which comply with a Forecast accepted by MYTV pursuant to Section F of this RAO.
- 2.9 Time for acceptance or rejection: MYTV must notify the Access Seeker that an Order is accepted or rejected within:
 - (a) the specified time frame in the Service Specific Obligations for the purpose of this

Paragraph; or

(b) the time frame within which it accepts or rejects equivalent Orders for itself,

whichever is shorter.

If MYTV notifies the Access Seeker that an Order is rejected, MYTV must advise the Access Seeker of the grounds of rejection and whether MYTV would be able to accept the Order in a modified form.

- 2.10 Notice of Acceptance: MYTV's Notice of Acceptance to the Access Seeker must contain the following information:
 - (a) the delivery date or activation date (as applicable), which must be the date that is requested by the Access Seeker, or, if that date cannot be met by MYTV, then no later than:
 - (i) the indicative delivery timeframe or activation timeframe i.e. twenty (20) Business Days, as specified in the Service Specific Obligations for the purpose of this Paragraph 2.7; or
 - (ii) the period of time taken by MYTV to deliver, or activate, such Services for itself,

whichever is shorter;

- (b) the date when civil works (if any) are intended to commence;
- (c) the charges applicable to fulfil the Order; including without limitation additional works such as internal wiring, right of way, land rental, local authority permits and third-party deposits;
- (d) such information as is reasonably necessary for the Access Seeker to benefit from access to the Services; and
- (e) the validity period, which shall be a period that is not shorter than three (3) months commencing from the date of the Notice of Acceptance ("Validity Period").
- (f) MYTV shall allow the Access Seeker a period of up to ten (10) Business Days after a request for additional information under Paragraph 2.6(b) above to provide MYTV with such information.
- 2.11 Commencement of delivery timeframes: The applicable delivery timeframe for an Order, as determined under Paragraph 2.7 (a) above, shall commence from:
 - (a) where the Access Seeker's confirmation of an Order is required under Paragraph 2.9 below, the date the Access Seeker confirms the Order in accordance with that Paragraph; and
 - (b) in any other case, from the start of the Validity Period.

2.12 Access Seeker's confirmation:

- (a) The Access Seeker's confirmation of an Order is not required if MYTV accepts the Order without change. A change may include circumstances where delivery dates are delayed, estimated charges are exceeded or any other matter that requires further confirmation from the Access Seeker before MYTV can proceed with the Order.
- (b) Where the Access Seeker's confirmation is required for MYTV to proceed with fulfilling an Order as provided for under Paragraph 2.9(a) above, MYTV shall permit the Access Seeker to provide its confirmation within the Validity Period and shall not provision the Order until the confirmation is received. Upon receipt of such confirmation, MYTV shall fulfil the Order in accordance with the Notice of Acceptance.
- 2.13 Estimated charges: If the Notice of Acceptance provided by MYTV contains estimates of charges (e.g. based on time and materials):
 - (a) MYTV shall not exceed the estimate without providing the Access Seeker with a written notice prior to exceeding the estimate that:
 - (i) the estimate will likely be exceeded;
 - (ii) an explanation of the reasons for exceeding the estimate; and
 - (iii) a further estimate of the charges for the work necessary to fulfil the Order;
 - (b) MYTV shall permit the Access Seeker to withdraw the Order without penalty within ten (10) Business Days of the notice given by MYTV under Paragraph 2.10(a) above if the revised estimate in that notice exceeds the original estimate by more than ten percent (10%);
 - (c) where the actual cost incurred by MYTV exceeds an estimate or revised estimate for a specific scope of work provided by MYTV due to:
 - (i) information or facts provided by the Access Seeker which are inaccurate or erroneous or not disclosed by the Access Seeker; or
 - (ii) a change in the scope of work by the Access Seeker,
 - the Access Seeker shall be obliged to pay MYTV for the actual cost incurred (but in no other circumstances); and.
 - (d) MYTV shall commence work after the Access Seeker confirms that it is agreeable to the estimate or revised estimate, whereby such confirmation is to be provided by the Access Seeker within the timeframe set out in Paragraphs 2.7(e) or 2.10(b) above, as applicable.
- 2.14 Reasons for rejection: MYTV may only reject an Order from an Access Seeker where:
 - (a) subject to Paragraph 6.2 of Section D of this RAO (as if references to 'Access Request' in that Paragraph were references to 'Order'), it is not technically feasible to provide

- access to the Services requested by the Access Seeker;
- (b) subject to compliance with Paragraphs 2.24 and 2.25 below, MYTV has insufficient capacity to provide the requested Services;
- (c) subject to Paragraph 2.13 below, the Order is in excess of the agreed Forecast levels;
- (d) the Order or variation request duplicates an Order awaiting fulfilment;
- (e) there are reasonable grounds to believe that the Access Seeker would fail to a material extent, to comply with the terms and conditions of the Access Agreement and such concern cannot be addressed to MYTV's satisfaction, acting reasonably (e.g. through the application of a security requirement in accordance with this RAO); or
- (f) there are reasonable grounds to believe that the Access Seeker would fail, in connection with the supply of the Services to protect the integrity of a Network, or the safety of individuals working on, or using services supplied by means of a Network or Equipment and such concern cannot be addressed to MYTV's satisfaction, acting reasonably (e.g. through the application of reasonable security or escorted access requirements).
- 2.15 Notice of rejection: MYTV's notice of rejection of an Order to the Access Seeker must:
 - (a) set out the grounds on which MYTV rejects the Order, at a sufficient level of detail to enable the Access Seeker to understand the basis of the rejection and to undertake its own re-assessment of the Order; and
 - (b) offer to meet, and meet if the offer is accepted by the Access Seeker, within five (5) Business Days of the notice of rejection of the Order to discuss the reasons for rejection and alternative methods of compliance.
- 2.16 Order in excess of Forecast: Notwithstanding Paragraph 2.11(b) above, MYTV must use its reasonable efforts to provide sufficient capacity to enable MYTV to accept and fulfil Orders from an Access Seeker for Services, which are in excess of the relevant Forecast. MYTV is only required to do so if, after meeting the Forecast requirements of other Access Seekers and itself, there is available capacity or MYTV could readily upgrade existing capacity. MYTV shall allocate the available capacity on a non-discriminatory basis to meet the over Forecast requirements of all Access Seekers and itself. MYTV is not required to supply Services in excess of the Forecast if, despite adopting any reasonable improvements (including upgrading capacity), this would cause a material degradation in the quality of Services provided to all Access Seekers and/or itself.
- 2.17 Indicative delivery times: The indicative delivery timeframe for DTBMS services is twenty (20) Business Days from the Notice of Acceptance or the confirmation of the order (as applicable) in accordance with Paragraph 2.9 above.
- 2.18 Early delivery dates: If MYTV, in the normal course of business, is able to offer a delivery date earlier than the delivery date that would otherwise apply, it must advise the Access Seeker in writing and, if requested by the Access Seeker, deliver access to the relevant Services at the earlier delivery date.

- 2.19 Delayed delivery dates: Where a delay in the delivery of an order is caused by the Access Seeker:
 - (a) MYTV shall notify the Access Seeker of the delay to the delivery date as soon as practicable after MYTV becomes aware of it;
 - (b) MYTV and Access Seeker must work together to minimise the delay; and
 - (c) the delivery date specified in the confirmed order or indicative delivery time set out above shall be extended for a further period as may be reasonably necessary by MYTV, and MYTV shall promptly notify the Access Seeker of the revised delivery date;

or, where there is a delay caused by MYTV or by a third party, that is not acting under MYTV's direction or control:

- (d) MYTV shall notify the Access Seeker of the delay to the delivery date, together with the reasons for the delay, as soon as practicable after MYTV becomes aware of the possible delay;
- (e) MYTV shall permit the Access Seeker to cancel the Order without penalty if the delay is longer than the equivalent time period for delivery of the Service; and
- (f) the delivery date shall be extended for a further period as may be reasonably necessary by MYTV, and MYTV shall promptly notify the Access Seeker of the revised delivery date.
- 2.20 Cancellation and variation of Orders: MYTV shall allow an Access Seeker to cancel or vary an Order at any time subject to Paragraph 2.18 below.
- 2.21 Cancellation or variation penalty: Except where this RAO provides that cancellation of an Order is to be at no penalty:
 - (a) MYTV may impose a charge for the cancellation or variation of the Order; and
 - (b) the charge which the Access Seeker is required to pay shall not exceed the lesser of the following amounts:
 - (i) the of costs necessarily incurred by MYTV which is directly attributable to the cancellation or variation; or
 - (ii) an amount equal to the sum of charges that would have been payable by the Access Seeker in the six (6) months immediately following the cancellation or variation had the Order not been cancelled or varied,

and reduced to the extent that those costs have been mitigated, or would have been mitigated had MYTV used its best endeavours to do so.

- 2.22 Testing and provisioning: MYTV shall:
 - (a) co-operate with the Access Seeker in relation to the testing and provisioning of ordered

- Services, including, but not limited to, by implementing a proof of concept if requested by the Access Seeker;
- (b) treat an Access Seeker's testing and provisioning on an equivalent basis to that which MYTV treats testing and provisioning for itself; and
- (c) may require reasonable co-operation by the Access Seeker in respect of such activities.

2.23 Resource charge: MYTV:

- (a) may charge the Access Seeker a one-off and recurring fees, to be determined by reference to the costs incurred by MYTV, for allocation of manpower and other resources to enable MYTV to test and fulfil an Order for new Services, provided that such one-off fee is justified by MYTV to the Access Seeker as necessary for MYTV to provide the requested Services; and
- (b) specifies the methodology and unit costs for fees under Paragraph 2.20(a) above can be referred under Paragraph 2, Part III of Section G, and in this RAO; and
- (c) may reasonably require the methodology and unit costs for fees under Paragraph 2.2(a) above that have not been included in this RAO be subject to a confidentiality agreement in accordance with Annexure 2 of this RAO.
- 2.24 Late delivery: If MYTV fails to meet the delivery date or any extended delivery date notified to the Access Seeker in accordance with Paragraph 2.16 (f) above, except where such failure has been caused solely by the Access Seeker's delay or a delay by a third party that is not acting under MYTV's direction or control (for example, where a local authority or landowner delays providing necessary approvals for works to commence), MYTV shall, without limitation to any other rights the Access Seeker may have under Part II of this RAO or the law, provide a rebate to the affected Access Seeker. The rebate shall be for an amount equivalent to the recurring charges payable for access to the Services for the period of MYTV's delay, and the methodology and unit rates for calculating such rebates shall be set out in MYTV's RAO. If MYTV alleges that a failure has been caused solely by the Access Seeker's delay or a delay by a third party not acting under MYTV's direction or control, MYTV shall have the burden of demonstrating:
 - (a) that allegation; and
 - (b) that MYTV has done all things reasonably practicable to minimise or avoid such failure.
- 2.25 Capacity Allocation Policy If the MYTV claims or is likely to claim that it has insufficient capacity to meet an Access Seeker's Forecasts or Orders, MYTV shall maintain a Capacity Allocation Policy, which:
 - (a) shall:
- i) be fair and reasonable;
- ii) be consistent, so far as practicable, with the MYTV's general duty of nondiscrimination in accordance with subsection 149(2) of the Act;

- iii) treat the requirements of all Access Seekers on an equivalent basis to the requirements of Access Provider's own divisions, subsidiaries, partners or other entities in which it has a direct or indirect equity, contractual or other interest; and
- iv) allocate the available capacity in the relevant Facilities and/or Services in proportion to each Operator's Forecast and/or Order requirements;

and

- (b) shall set out the MYTV's plans to expand their capacity over time (if any), where such information must be provided to Access Seekers on a non-discriminatory basis in terms of its content and frequency of updates
- 2.26 Acceptance on queue: MYTV shall promptly notify an Access Seeker at the time of providing an acknowledgement of receipt of the Order under Paragraph 2.5 above (and as specified in the Notice of Receipt under Paragraph 2.6 above), of their acceptance of, and position in, MYTV's queue.
- 2.27 Constrained capacity: If MYTV reasonably believes that the capacity in any Services required by:
 - (a) the Access Seeker pursuant to the relevant Forecast and/or Order;
 - (b) other Access Seekers, pursuant to their relevant Forecasts and/or Orders; and
 - (c) MYTV, for the purposes of its own divisions, subsidiaries, partners or other entities in which it has a direct or indirect equity, contractual or other interest,

would, in aggregate, exceed the capacity which MYTV will be in a position to be able to provide, MYTV must:

- (d) notify all Access Seekers to whom relevant capacity is supplied; and
- (e) allocate the available capacity between itself, the Access Seeker and other Access Seekers in accordance with MYTV's Capacity Allocation Policy.
- 2.28 Capacity Allocation Policy: MYTV's capacity allocation will be on a first come first serve basis.
 - (a) MYTV shall maintain a Capacity Allocation Policy, shall be:
 - (i) fair and reasonable;
 - (ii) consistent, so far as practicable, with MYTV's general duty of non-discrimination in accordance with Section 149(2) of the Act;
 - (iii) treat the requirements of all Access Seekers on an equivalent basis to the requirements of MYTV's own divisions, subsidiaries, partners or other entities in which it has a direct or indirect equity, contractual or other interest; and
 - (iv) allocate the available capacity in the relevant Services in proportion to each Operator's Forecast and/or Order requirements; and
 - (b) MYTV shall set out its plans to expand their capacity over time (if any), where such

information must be provided to Access Seekers on a non-discriminatory basis in terms of its content and frequency of updates.

2.29 All orders and/or related information shall be addressed to:

MYTV Broadcasting Sdn. Bhd. 2271, Jalan Usahawan 2, Cyber 6, 63000, Cyberjaya, Selangor

Attention: Group Head Regulatory

Email: regulatory@mtvb.my

PART III - DECOMMISSIONING OBLIGATIONS

1. General

- 1.1 Part IV of Section F sets out the decommissioning obligations that are applicable only in relation to the provision of Services listed in the MYTV RAO preface.
- 1.2 Decommissioning notice: Except where MYTV is required to vacate a site where a Point of Interface is located, or any other Service which relies on MYTV's use of that site, as a result of a third party landlord's notice (under an arm's length tenancy agreement) or a local authority's notice, MYTV must provide no less than:
 - (a) one (1) year's notice in writing to all relevant Access Seekers prior to the decommissioning of a Point of Interface; or
 - (b) six (6) months' notice in writing to all relevant Access Seekers prior to the decommissioning of any Services which rely on MYTV's use of that site.

Where MYTV is required to vacate the site as a result of a third party landlord's notice (under an arm's length tenancy agreement) or a local authority's notice, MYTV must provide all relevant Access Seekers with as much notice as possible in relation to the matters in Paragraph 1.2 (a) and 1.2 (b).

- 1.3 Alternative arrangements: MYTV which notifies an Access Seeker of its intention:
 - (a) to decommission a Point of Interface, shall provide to the Access Seeker a functionally equivalent interconnection at another Point of Interface on terms and conditions and at a recurring charge which are not disadvantageous to the Access Seeker, relative to the terms and conditions and recurring charge applicable in respect of the Point of Interface that is proposed to be decommissioned, for a period that is not less than three (3) years from the date of decommissioning; or
 - (b) to decommission any other Services, shall provide to the Access Seeker access to alternative Services on terms and conditions and at a recurring charge which are not disadvantageous to the Access Seeker, relative to the terms and conditions and

recurring charge applicable in respect of the Services that are proposed to be decommissioned, for a period that is not less than three (3) years from the date of decommissioning.

- 1.4 Decommissioned Point of Interface compensation: MYTV shall pay the Access Seeker reasonable costs, necessarily incurred in:
 - (c) decommissioning any links to the Point of Interface that is proposed to be decommissioned, that are, or will be, rendered redundant by the proposed decommissioning;
 - (d) installing or otherwise procuring links between the Point of Interface that is proposed to be decommissioned and the substitute Point of Interface to be provided pursuant to paragraph 1.3(a) above;
 - (e) the carriage of traffic between the Point of Interface that is proposed to be decommissioned and the subsubstitute Point of Interface to be provided pursuant to paragraph 1.3(a) above for a period that is not less than three (3) years from the date of decommissioning.
- 1.5 Decomissioned Services compensation: Except where decommissioning is caused by Force Majeure, MYTV shall pay the Access Seeker's reasonable costs, necessarily incurred in rearranging Equipment to connect to alternative Services offered in accordance with paragraph 1.3 (b) above.

PART IV - BILLING AND SETTLEMENT OBLIGATIONS

- 1.1 The Access Seeker shall pay MYTV the charges for the relevant Services supplied by MYTV to the Access Seeker, as specified in this RAO.
- 1.2 The Parties shall bear and pay all taxes as required by Malaysian law that result from the implementation of this RAO.
- 1.3 All payments must:
 - (a) be paid on the due date unless otherwise agreed in writing by the Parties;
 - (b) be paid by electronic transfer to MYTV or exceptionally, by cheque to the nominated account(s) of MYTV if agreed by MYTV; and
 - (c) must be accompanied by such information as is reasonably required by MYTV to properly allocate payments received.
- 1.4 No set-off: Unless otherwise agreed between MYTV and Access Seeker in an Access Agreement, MYTV may not set-off Invoices except where the Access Seeker is in liquidation or at least three (3) Invoices have been issued and such Invoices have not been paid (excluding disputed amounts)

- 1.5 For the purpose of clarification, the Security Sum does not relieve the Access Seeker from its obligations to pay amounts of any invoices to MYTV's right to suspend, disconnect, or terminate the relevant network facilities or network services due to non-payment of any sums due or payable to MYTV and shall not be subject to any deduction or set-off unless otherwise agreed in writing by MYTV.
- 1.6 (a) MYTV shall be entitled to revise the Security Sum a maximum of once in any twelve (12) months period in any of the following event:-
 - (i) upon the provisioning of new network services to the Access Seeker; or
 - (ii) where there is material change in circumstances in relation to the Access Seeker's creditworthiness. For clarification, a material change in circumstances includes, but is not limited to, a failure by the Access Seeker to pay on the due dates at least three (3) invoices (so long as those amounts have not been disputed in good faith). If the amounts in the invoices are disputed in good faith, this will not constitute a material change in circumstances for purposes of this Paragraph 1.5 (a) (iii).
 - (b) Where the Security Sum is revised pursuant to Paragraph 1.5 (a) above, the Access Seeker shall within ten (10) Business Days from the written request of MYTV, deposit the new Security Sum with MYTV.
 - (c) Where the Access Seeker deposits monies in lieu of a Bank Guarantee, such monies shall be deposited in a separate interest bearing account ("said accounts") and any interest secured thereon be held by MYTV in addition to the Security Sum. MYTV shall forward to the Access Seeker a statement of the said accounts annually.
- 1.7 MYTV shall allow an Access Seeker to dispute any amount in an invoice if, the Access Seeker notifies MYTV within thirty (30) Business Days after the date of receipt of such Invoice.
- 1.8 (a) In the event MYTV elects to suspend or terminate the provisioning of relevant Access Service(s) to the Access Seeker, MYTV shall have the right to use the Security Sum (together with any interest thereon) to set off any outstanding sum due and payable to MYTV by the Access Seeker.
 - (b) Subject to Paragraph 1.7 (a) above, upon expiry of the Access Agreement, the Security Sum deposited with MYTV or parts thereof, together with interest thereon, (if any) shall be returned and/or refunded to the Access Seeker.
- 1.8 Invoices: MYTV shall use its best endeavours to issue to the Access Seeker an Invoice in writing or in electronic form (as requested by the Access Seeker) within one (1) month of the end of each Billing Cycle in accordance with Paragraph 1.10 below for amounts due in respect of the supply of Services during the relevant Billing Period.
- 1.9 Currency: Unless otherwise agreed by MYTV and Access Seeker in an Access Agreement, MYTV shall state all Invoices in Ringgit Malaysia and payment shall be made by the Access Seeker in Ringgit Malaysia.
- 1.10 Billing Cycle: MYTV shall issue Invoices in accordance with the Billing Cycles specified in the

- Service Specific Obligation, Section G, Part I Paragraph 10 and Part II Paragraph 4, except where a different Billing Cycle is agreed with the Access Seeker in an Access Agreement.
- 1.11 Billing verification information: MYTV shall provide, with each Invoice, such information as may be reasonably necessary for the Access Seeker to verify rates and charges contained in an Invoice.
- 1.12 Summarised Invoice and billing information: MYTV shall provide the Access Seeker, on written request, with an aggregated summary of billings for access to the Services provided to the Access Seeker, in monthly tranches.
- 1.13 Billing error: If an Operator discovers an error in an Invoice, it must promptly notify the other Operator. The Operator which made the error must make necessary adjustments to correct that error within one (1) month of notification.
- 1.14 Time for payment: Subject to Paragraph 1.15 below, MYTV shall allow an Access Seeker no less than one (1) month from the date of receipt of an Invoice for the Access Seeker to make the payment. This Paragraph 1.14 should not be construed as preventing MYTV from granting a discount to an Access Seeker as an incentive to make early payments.
- 1.15 Withholding of disputed amounts: MYTV shall allow an Access Seeker to withhold payment of any amount disputed in good faith by the Access Seeker if:
 - (a) the Access Seeker notifies MYTV within fifteen (15) Business Days from the date of receipt of the Invoice of such dispute (unless otherwise agreed by MYTV and Access Seeker in an Access Agreement); and
 - (b) the Access Seeker's notification specifies the information referred to in Paragraph 1.17 below.
- 1.16 Billing Disputes: MYTV shall allow an Access Seeker to dispute any amount in an Invoice and notifies MYTV within thirty (30) Business Days after the date of receipt of such Invoice.
- 1.17 Notification of Billing Dispute: MYTV may require an Access Seeker to provide the following information when disputing any amount in an Invoice:
 - (a) the reasons for which the Invoice is disputed;
 - (b) the amount in dispute;
 - (c) details required to identify the relevant Invoice and charges in dispute including:
 - (i) the account number;
 - (ii) the Invoice reference number;
 - (iii) the Invoice date;
 - (iv) the Invoice amount; and

- (v) billing verification information; and
- (d) evidence in the form of a report, indicating anyrelevant traffic data which is in dispute.
- 1.18 Billing Dispute resolution: MYTV and an Access Seeker must comply with the Dispute Resolution Procedures applicable to Billing Disputes.
- 1.19 Interest: Except for any amount in an Invoice being disputed by an Access Seeker in good faith in accordance with Paragraph 1.16 above, MYTV may charge interest on any amount outstanding from an Access Seeker from time to time, in respect of that overdue sum for the period beginning on its due date and ending on the date of the receipt of the overdue sum by MYTV. The interest that may be charged by MYTV shall be at the rate of two percent (2%) per annum above Malayan Banking Berhad's base rate calculated daily from the due date until the date of actual payment. Payments which are overdue by more than two (2) months will bear interest at the rate of three percent (3%) per annum above Malayan Banking Berhad's base rate calculated from the due date until the date of receipt by MYTV of full payment. For clarification, MYTV shall not charge interest on an amount which is disputed by an Access Seeker in good faith.
- 1.20 Backbilling: Unless otherwise agreed by MYTV and Access Seeker in an Access Agreement, MYTV may include omitted or miscalculated charges from an earlier Invoice in a later Invoice, or issue an Invoice for charges which have previously not been invoiced provided that MYTV is able to substantiate the charges to the Access Seeker and such inclusion, amendment or issuance is made within three (3) months from the end of the Billing Cycle in which the calls were made or in which other Services were provided.

PART V - OPERATION AND MAINTENANCE OBLIGATIONS

- 1. General
- 1.1 Part VII of Section F is applicable only in relation to the provision of Services listed in the MYTV RAO preface.
- 1.2 Where applicable, the Parties will:
 - (a) Use their reasonable endeavours within four (4) weeks from the commencement date to negotiate, agree and document as soon as reasonably practicable the new or modified provisions of the Operations and Maintenance Manual and any other Manuals which the Parties deem necessary to establish pursuant to the Access Agreement;
 - (b) Comply with the operational procedures and methods set out in the Manuals; and
 - (c) Where such procedures and methods have not been agreed, negotiate operational procedures and methods, in relation to:
 - (1) the planning, ordering, provisioning and delivery of the relevant network facilities or network services;

- (2) the management of the relevant network facilities or network services including:
 - Quality of service indicators, reporting on performance in terms of those indicators and determining the appropriate action to be taken in the event that Service quality falls below the agreed indicator levels;
 - (ii) Network operations in the event of Network failure, congestion and blockage and ensuring that the Parties' Network are adequately protected from harm;
 - (iii) Test procedures and other technical and operational matters relating to the provision of network facilities or network services by MYTV to the Access Seeker;
 - (iv) The handling of Customer operations; and
 - (v) Such other matter as MYTV determines.
- 1.3 Fault reporting service: Each Operator shall be responsible for the operations and maintenance of its own facilities and services. Each Operator shall establish and maintain a fault reporting service that allows Customers who are directly connected to the Network of that Operator relating to any Network, Facility and/or Service.
- 1.4 Rectification target times: SOC of DTBMS is a 24-7 proactive systems monitoring, management and control operations. SOC is also the single point of contact for all technical support for the Access Seeker, where SOC shall escalate the fault accordingly and record on findings during troubleshoot process.
- 1.5 MYTV technical support contact details are as per table below:

MYTV SOC Email	soc@mtvb.my
MYTV SOC Telephone Number	03-8313 6916

- 1.6 Cross-referrals: If a Customer reports a fault to an Operator:
 - (a) when the Customer is directly connected to another Operator; or
 - (b) which clearly relates to a Network, Facility and/or Service of another Operator,

the Operator which receives the report shall promptly inform the other Operator of the reported fault, or refer that Customer to the other Operator's fault reporting service.

1.7 Network fault responsibility: The Operator in whose Network the fault occurs is responsible for rectifying it and restoring services including for the purpose of restoring the supply of Facilities and/or Services which are used in another Operator's Network.

- 1.8 Faults affecting other Networks or Equipment: If an Operator identifies a fault occurring in its Network or with its network facilities which may have an adverse effect on the other another Operator's Network, network facilities, network services or Equipment, the first-mentioned Operator must promptly inform the other Operator of:
 - (a) the existence of the fault;
 - (b) the actions being taken by the first mentioned Operator to rectify the identified faults and restore the service; and
 - (c) the outcome of those actions.
- 1.9 Bear own costs: Each Operator is responsible for establishing and maintaining a fault reporting service at its own cost irrespective of the location of the fault.
- 1.10 Fault rectification: Each Operator shall rectify faults on an equivalent and non-discriminatory basis.
- 1.11 Each Operator shall respond to and rectify faults within the lesser of:
 - (a) timeframes set out in a relevant Service Specific Obligation;
 - (b) timeframes which will result in compliance by all affected Operators with any applicable mandatory standards that apply to service availability and restoration; and
 - (c) timeframes equivalent to that which MYTV provides to itself.

MYTV shall respond to and rectify identified faults of its Network and Service based on the maximum allowable time as defined in the table below:

Service Item	Fault Rectification Timeframes
Service Operation Centre	7 Days * 24 Hours
Response Time from Notification	30 mins
Service Rectification Time (Fault Related)	24 hours - 7 days
Permanent Resolution (Non-Bug) from	20 Business Days
Notification	
Permanent Resolution Time (Bug)	90 Calendar Days
Access and Rectification at External	Within 7 days after receiving notification.
Transmission Sites	

- 1.12 Complaints handling: The Operators must report all interconnection and access outages that relate to Networks and Services to MYTV's Service Operation Centre.
- 1.13 Planned maintenance: If an Operator intends to undertake planned maintenance ("Maintenance Operator") which may affect an Access Seeker's Network, Facilities and/or Services, the Maintenance Operator must:
 - (a) provide at least the greater of the time which it notifies its own Customers and ten (10)

Business Days' notice of the planned maintenance;

- (b) use its reasonable endeavours to minimise any disruption to the carriage of communications that crosses or would cross both Operators' Networks, and which are caused by the maintenance or re-routing; and
- (c) where the Operators agree that it is practicable, provide alternative routing or carriage at no additional cost to the Access Seeker.
- 1.14 Emergency maintenance: If a Maintenance Operator needs to undertake emergency maintenance which may affect the other Operator's Network, the Maintenance Operator must. if it is able to:
 - (a) provide at least twenty-four (24) hours' notice of the planned maintenance;
 - (b) use its reasonable endeavours to minimise any disruption to the carriage of communications that crosses or would cross both Operators' Networks, and which are caused by the maintenance or rerouting; and
 - (c) where the Operators agree that it is practicable, provide alternative routing or carriage at no additional cost to the other Operator.

PART VI - TECHNICAL OBLIGATIONS

- 1.1 Compliance: Operators shall adhere to the relevant guidelines issued by the Commission from time to time to the extent that they have not been expressly revoked and are not inconsistent with any technical obligations set out in this RAO.
- 1.2 Prevention of technical harm: An Operator must take reasonable measures to ensure that the interconnection and access does not cause physical or technical harm to the other Operator's Network, which measures shall be no less robust than the measures which the Operator takes in respect of new facilities or Equipment incorporated into its own Network.
- 1.3 Technical Standards: An Operator must comply with any applicable technical Standard adopted by the Commission under Chapter 3 of Part VII of the Act.
- 1.4 No Interference: An Operator must not do anything, or knowingly permit any third person to do anything, in relation to a Network, network facilities, network services or Equipment which:
 - (a) causes interference; or
 - (b) materially obstructs, interrupts or impedes the continuous use or operation of, the Network, network facilities, network services or Equipment of another Operator.
- 1.5 Notice of interference and rectification: If an Operator (Notifying Operator) notifies another Operator that the other Operator's Network, network facilities, network services or Equipment is causing interference to the Notifying Operator's Network, network facilities, network services or Equipment:

- (a) the other Operator shall rectify the situation as soon as possible and, in any case, so that no interference is caused within twenty-four (24) hours of receiving notice from the Notifying Operator, so that no interference is caused; or
- (b) if the other Operator is not able to locate the source of the interference within twenty-four (24) hours under Paragraph 1.5(a) above, the other Operator shall promptly notify the Notifying Operator, and both Operators shall meet as soon as possible, and in any case, within twenty-four (24) hours of such notice and jointly examine each other's Network, network facilities, network services or Equipment to locate the source of the interference.

PART VII - TERM, SUSPENSION AND TERMINATION OBLIGATIONS

1.1 Term: The Operators shall, unless otherwise required by the Access Seeker, enter into an Access Agreement with a term of no less than five (5) years from the date of execution of the Access Agreement.

1.2 Termination circumstances:

- (a) Subject to Paragraph 1.6 below, MYTV may only terminate an Access Agreement, whether in whole or in part, if any of the circumstances referred to in Paragraph 1.2 (a)(i), 1.2 (a)(ii) or 1.2 (a)(iii) below apply and MYTV has notified the Access Seeker that it will terminate where:-
- (i) The Access Seeker has materially breached the Access Agreement and MYTV has notified the Access Seeker that it will terminate the said Agreement in no less than thirty (30) days if the Access Seeker has not remedied its breach by the end of that period and the Access Seeker has failed to remedy its breach in accordance to such a notification;
- (ii) the Access Seeker is subject to a winding up order; or
- (iii) a Force Majeure event has continued for a period of more than ninety (90) days.

MYTV shall forward to the Commission copy of the notice of termination at the same time as providing the notice of termination to the Access Seeker.

- (b) Except where permitted under paragraph 1.5 below, an Access Seeker may only terminate an Access Agreement, whether in whole or in part (for example, only to the extent relating to a particular Facility or Service, or at a particular site), if any of the circumstances referred to in Paragraph 1.2(b)(i), 1.2(b)(ii) or 1.2(b)(iii) below apply, and the Access Seeker has notified the Access Provider that it will terminate where:
- (i) MYTV has materially breached the Access Agreement and the Access Seeker has notified MYTV that it will terminate the said Agreement in no less than thirty (30) days MYTV has not remedied its breach by the end of that period and MYTV has failed to remedy its breach in accordance to such a notification;

- (ii) MYTV is subject to a winding up order; or
- (iii) A Force Majeure event has continued for a period of more than ninety (90) days.
- 1.3 Changes in law: Where the continued operation of the Access Agreement or access to any Access Service(s) provided by MYTV is or will be unlawful (as a result of a legislative change), the Access Seeker and MYTV shall meet within five (5) Business Days of becoming aware of the relevant change in law to review whether access to the relevant Access Service(s) may be provided by MYTV on different terms and conditions (which are acceptable to the Access Seeker). If the parties cannot agree to the provision of access on different terms and conditions, either Party may request resolution of the dispute in accordance with Dispute Resolution Procedures.
- 1.4 Suspension circumstances: Subject to Paragraph 1.6 below, MYTV may only suspend access to any Access Service(s) whether in whole or in part, in the following circumstances:
 - (a) The Access Seeker's Facilities materially and adversely affect the normal operation of MYTV's Network or are a threat to any person's safety;
 - (b) The Access Seeker's Facilities cause material physical or technical harm to any Facilities of MYTV or any other person;
 - (c) Where the Access Seeker has failed to pay invoices in accordance with Part VI of Section F of this RAO, and has failed to rectify such non-compliance within thirty (30) days of receiving notice from MYTV (and subject to any right that the Access Seeker has under Part VI, Section F of this RAO to dispute any amount in an Invoice);
 - (d) Where the Access Seeker has failed to provide the new security amount as required under Section 5.3.9, 5.16.7 and 5.16.8 of the MSA determination.
 - (e) Where force majeure provision applies; or
 - (f) The Access Seeker breaches any laws, regulations, rules, or standards which has a material adverse effect on MYTV or the provision by MYTV of Access Service(s) under the Access Agreement.

For the purposes of this Paragraph 1.4, MYTV must provide Access Seeker five (5) Business Days' notice in writing, including written reasons, prior to suspending access to any Access Services(s).

1.5 Term of supply: Unless otherwise agreed by the Access Provider and Access Seeker in an Access Agreement, and subject to the Access Provider not being able to provide access as a result of Force Majeure, the Access Provider shall only require the Access Seeker to acquire access to Facility and/or Service under the Access Agreement for a minimum period as follows:

Facilities and/or Services	Minimum term	
Network facilities access	Three (3) years	

- 1.6 Notice: Prior to terminating, suspending, or seeking to materially vary an Access Agreement or access to any Services provided under it, MYTV must notify the Commission in writing of the proposed action to be taken and the reasons why it considers such action is appropriate. The Commission may invite any affected Access Seeker to make submissions to the Commission regarding the proposed termination, suspension or material variation. MYTV:
 - (a) shall only give effect to the proposed termination, suspension or material variation with the Commission's written consent and subject to any time delay or conditions which the Commission may specify (if any). The Commission will endeavour to respond to MYTV's notice within ten (10) Business Days or such other period that the Commission considers is reasonable;
 - (b) must not give effect to the proposed termination, suspension or material variation unless MYTV has received written consent from the Commission to such termination, suspension or material variation; and
 - (c) shall take all steps practicable to minimise disruptions and inconvenience to the Customers of the Access Seeker, including providing the Access Seeker with a reasonable period to make alternative arrangements prior to the suspension or termination of the Access Agreement, or access to Services provided under it.
- 1.7 Undertakings: If the Operators to an Access Agreement adopt the terms and conditions specified in an undertaking that has been registered with the Commission in accordance with the Act, the Operators must notify Commission within five (5) Business Days of such adoption. In such circumstances, the terms and conditions of the Access Agreement will continue to be in force for the remainder of the term of that Access Agreement, even if the access undertaking is withdrawn or expires prior to the expiry of that term.
- 1.8 Post-termination fees: MYTV shall not recover any additional charges, costs or expenses on termination or suspension of an Access Agreement or access to any Access Service(s) provided under it except:
 - (a) Charges invoiced in arrears and not yet paid; or
 - (b) Compensation for the remaining unexpired period.

Such charges must be reduced to reflect any cost savings to MYTV from not having to supply the Services to the extent that they have been suspended or terminated and MYTV must use reasonable endeavours to mitigate its costs of termination or suspension and maximise cost savings.

- 1.9 Upfront charges refund: On termination of an Access Agreement or access to any Access Service(s) provided under it, MYTV shall refund to the Access Seeker all amounts paid in advance to the extent that the amount (or part of the amount calculated on pro-rata basis) relate to the period after the date of effect of such termination.
- 1.10 Deposits and guarantees: Notwithstanding the obligation in Paragraph 1.9 above, MYTV shall:

- (a) Within sixty (60) days of termination of the Access Agreement, refund to the Access Seeker any deposit paid (without interest) provided all other amounts due and payable by the Access Seeker to MYTV have been paid; and
- (b) immediately upon termination of the Access Agreement, unconditionally waive any rights under the guarantees provided by the Access Seeker except in respect of amounts due and payable by the Access Seeker to MYTV as at the date of termination.
- 1.11 Notice: Prior to terminating, suspending, or seeking to materially vary a MYTV Access Agreement (including any part thereof) or access to any Facilities and/or Services provided under it, MYTV must notify the Commission in writing of the action MYTV proposes to take and the reasons why it considers such action is appropriate. The Commission may invite any affected Access Seeker to make submissions to the Commission regarding the proposed termination, suspension, or material variation. MYTV:
 - a. shall only give effect to the proposed termination, suspension, or material variation with the Commission's written consent and subject to any time delay or conditions which the Commission may specify (if any). The Commission will endeavor to respond to MYTV's notice within ten (10) Business Days or such other period that the Commission considers is reasonable;
 - b. must not give effect to the proposed termination, suspension, or material variation unless MYTV has received written consent from the Commission to such termination, suspension, or material variation; and
 - c. shall take all steps practicable to minimize disruptions and inconvenience to the Customers of the Access Seeker, including providing the Access Seeker with a reasonable period to make alternative arrangements prior to the suspension or termination of the MYTV Access Agreement or access to Facilities and/or Services provided under it.

PART VIII - LEGAL BOILERPLATE OBLIGATIONS

1. General

Part X of Section F sets out the legal boilerplate procedures that may be applicable as set out in Section 5.16 Legal Boilerplate Obligations of MSA Determination with exception to Section 5.16.3 and 5.16.9.

- 1.1 Mutual compensation: Each Operator must establish mutually acceptable compensation arrangements with each other Operator (including bill-and-keep arrangements).
- 1.2 Equal representatives: Each Operator must appoint an equal number of representatives to an Interconnect Steering Group (and such other working groups as may be agreed upon) to manage the smooth and timely implementation of the terms and conditions of Access Agreements or Dispute Resolution Procedures, as applicable.
- 1.3 Complete charges: MYTV shall specify all charges in an Access Agreement and shall not attempt to recover any other costs, expenses or charges which are not specified in the Access Agreement except where such work is to be done on a time and materials basis in which case MYTV shall do such work in accordance with a quotation agreed with the Access Seeker as set

out in this RAO.

- 1.4 Intellectual Property: Each Operator shall licence to other Operator under an Access Agreement on a royalty-free basis, all Intellectual Property rights necessary for the ongoing operation of the Access Agreement and the inter-operability of the Operators' Networks, subject to any relevant third party licences. The term of the licence must be consistent with the term of the relevant Access Agreement.
- 1.5 Review: MYTV shall specify in an Access Agreement prepared that such Access Agreement shall be reviewed:
 - (a) If the Minister issues a direction or determination relating to its subject matter;
 - (b) If the Commission issues a direction or determination relating to its subject matter;
 - (c) If the Act or the MSA is amended in relation to its subject matter;
 - (d) By agreement of each of the parties; or
 - (e) If a condition of the Operator's licence is amended or deleted or a new condition is imposed in relation to its subject matter.
- 1.6 Costs and expenses: Each Operator shall bear its own costs and expenses in relation to the preparation, negotiation and execution of an Access Agreement to which they are parties. The cost for stamping of the said Access Agreement shall be borne by the Access Seeker.
- 1.7 Applicable laws: An Operator shall include a provision in all Access Agreement prepared by it which provides that the Access Agreement will be governed by the laws of Malaysia and that Operators will comply with all applicable directions issued by the Malaysian regulatory authorities.

SECTION G: SERVICE SPECIFIC OBLIGATIONS

PART I - DIGITAL TERRESTRIAL BROADCASTING MULTIPLEXING SERVICE (DTBMS) AND DIRECT-TO-HOME (DTH)

- 1. Digital Terrestrial Broadcasting Multiplexing Service ("DTBMS") consist of the following services:
- 1.1 Standard Definition Television Channel ("SD");
- 1.2 High Definition Television Channel ("HD"); and'
- 1.3 Digital Radio Channel ("Radio").
- 2. Generally, the terms and conditions which are applicable to the DTBMS are as follows:
- 2.1 Standard Definition Television Channel
 - (a) Standard Definition Television Channel is a television broadcasting channel delivered over DTBMS with a standard bit rate allocation of 2 Mbps inclusive of single pair of

stereo (dual channel) of associated audio, single subtitling language and single interactive service signalling, without conditional access information.

- (b) The configuration and parameter of SD Channel can be any or combination of the following specification:
 - MPEG-4 AVC H.264 encoding with video resolution of 576i/25, Frame Rate of 25Hz
 @Field Rate of 50Hz;
 - Interlaced and aspect ratio of 4:3;
 - Audio encoding with HE-AAC v2 stereo audio coding;
 - Input interface physical connection : BNC; and
 - Input signal format : SDI.

2.2 High Definition Television Channel

- (a) High Definition Television Channel is a television broadcasting channel delivered over DTBMS with a standard bit rate allocation of 4 Mbps inclusive of single pair of stereo (dual channel) of associated audio, single subtitling language and single interactive service signalling, without conditional access information.
- (b) The configuration and parameter of HD Channel can be any or combination of the following specification:
 - MPEG-4 AVC H.264 encoding with video resolution of 1080i/25, Frame Rate of 25Hz
 @Field Rate of 50Hz;
 - Interlaced or video resolution of 720p/50, Frame Rate of 50Hz @Field Rate of 50Hz, Progressive, and Aspect Ratio of 16:9;
 - (Optional) Audio encoding with E-AC-3 streams (including AC-3) up to 5.1 multi-channel coding or MEPG-4 HE-AAC level 4 (including AAC-LC) up to 5.1 multichannel coding;
 - Input interface physical connection : BNC; and
 - Input signal format : SDI.

2.3 Digital Radio Channel

- (a) Radio broadcasting channel delivered over DTBMS with a standard bit rate allocation of 50 Kbps for a single pair of stereo (dual channel) audio.
- (b) The configuration and parameter of Radio Channel can be any or combination of the following specification:
 - Audio encoding HE-AAC v2 coding;

- Input interface physical connection : BNC; and
- Input signal format : SDI.
- 3. Bit rate allocation: The Access Seeker should have the flexibility to request for a bit rate allocation that is different to the ones specified by MYTV and MYTV must reasonably provide such bit rate unless it is technically infeasible for MYTV to do so.
- 4. Compression: The Access Seeker should be allowed to specify digital compression and decompression technology reasonably required to deliver its services and MYTV should comply with the Access Seeker's request to the extent possible.
- 5. MYTV shall only apply conditional access to an Access Seeker's transport stream if specifically requested to do so by Access Seeker.
- 6. Acknowledgement of receipt: For the purposes of Paragraph 2.5 of Section F Part II of this RAO, MYTV shall acknowledge receipt of each Order for DTBMS within two (2) Business Days.
- 7. Time for acceptance or rejection: Subject to any shorter timeframe required under Paragraph 2.15 of Section F Part I of this RAO, MYTV must notify an Access Seeker that an Order for DTBMS is accepted or rejected within ten (10) Business Days after issuing the Notice of Receipt in respect of the Order.
- 8. Indicative delivery timeframe: For the purposes of Paragraph 2.14 of Section F Part II of this RAO, the indicative delivery timeframe for DTBMS is twenty (20) Business Days. For clarification, the indicative delivery timeframe in this Paragraph commences from the Notice of Acceptance or confirmation of the Order (as applicable) in accordance with Paragraph 2.9 of Section F Part II of this RAO.
- 9. Billing Cycle: For the purposes of Paragraph 1.10 of Section F Part VI of this RAO, between the Operators, the Billing Cycle for DTBMS will be monthly.
- 10. Redundancy: MYTV shall provide redundancy in respect of the DTBMS to ensure availability of the Access Seeker's service at all times.
- 11. Technical standards: An Operator must comply with the Commission Determination on the Mandatory Standard for Free to Air Transmission of Digital Terrestrial Television Service, Determination No. 1 of 2011 including such modification or variation and any other mandatory standards as may be determined by the Commission from time to time.
- 12. Service Availability (SA): MYTV shall provide access to the DTBMS and DTH to Access Seekers within the thresholds stated in Table A and items as follows:

Parameter	Quarterly threshold (%)	
SA for DTBMS	99.0% per site, per transmitter	
SA for DTH	99.0% per DTH uplink transmission	

Table A

Note: Two additional parameters described in (a) and (b) below contribute to the determination of DTMBS service availability, while (c) determines DTH service availability.

- a) Modulation Error Ratio (MER): MYTV shall ensure to maintain the MER at all its DTBMS transmitters to be remained at a minimum of 31dB at the combiner output consistently. For the avoidance of doubt, the MER measurement this paragraph shall be collected at Physical Layer Pipe (PLP) (rms) or L1 (rms), which its measurement is dependent on the capability of the probe.
- b) Transmitter Output Power (TOP): MYTV shall ensure to maintain the TOP at all its DTBMS transmitters to be consistently maintain a minimum efficiency of no more than half from the specified TOP in the approved Apparatus Assignment by the Commission. The TOP for each transmitter at all sites is specified under the 'List of Transmitter Output Power (TOP)' stated in the Access Agreement
- c) Direct-to-home (DTH) Transmission: MYTV shall ensure to maintain the DTH transmission to be consistently maintain at a minimum efficiency. The efficiency is determined by the availability of all components involved in the DTH transmission, collectively contributing to the overall availability of the DTH service.
- 13. Service Credit: MYTV and Access Seeker shall discuss and negotiate in good faith the provision of service credit in which any DTBMS and DTH services provided by MYTV does not meet the relevant requirements as specified in the Paragraph 12 and 15 of this RAO or any other agreement as agreed between Access Seeker and MYTV, other than to extent to which non-compliance arises from:
 - (a) any act or omission by the Access Seeker other than in accordance to the direction of MYTV:
 - (b) force Majeure; or
 - (c) any act and/or omission and/or negligence by any third party whom not under any direction of MYTV.
- 14. Amount of service credit: The amount of any service credit for the purposes of Paragraph 12, and 15 of this section shall be calculated based on the following formula or any other formula as agreed between Access Seeker and MYTV:

Service credit per site = $T \times P \times D$

T = per minute tariff, referred to the pricing as agreed by both parties

- P = the proportion of the Malaysian population affected by the site downtime of any particular site. For number of population of DTBMS and DTH service, MYTV shall make reference to the population listed under Annexure 4.
- D = downtime in minutes, that the site is unavailable to provide service or below the agreed service availability for MER or TOP.

The maximum cap of the service credit amount shall be a sum mutually agreed upon between MYTV and Access Seeker.

15. Service Restoration: MYTV shall at best to respond to and restore identified downtime occurrence of its Network and Service based on the maximum allowable time as defined in the table below:

Category	Restoration time	No. of sites
Α	≤ 4 hours	33
В	≤8 hours	11
С	≤8 hours	3
	(for incident that happens between 0800 to 1900)	
	≤ 8 hours from 0800	
	(for incident that happens between 1900 to 0800)	

Table B

The list of the sites for each category is specified under the 'List of Transmitter Sites' in Annexure 4.

- 16. As a result of previous Force Majeure in the adjacent area to Telapak Burok site, there is a limited access to the site which may affect the restoration time by MYTV in any case of site downtime occurrence. The Telapak Burok site is provisionally listed under Category B and the status quo of the site shall remain until the access to the site is considered safe by MYTV to be listed under Category A.
- 17. Service restoration reporting: Should MYTV failed to restore the identified downtime occurrence within the specified restoration time in Table B above, MYTV must submit a comprehensive report within seven (7) working days, outlining necessary justification to the Commission.
- 18. Service restoration service credit: In any situation that service restoration falls beyond the restoration time in Table B or any other restoration time agreed by the parties, MYTV and Access Seeker shall discuss and negotiate in good faith for the provision of service credit.
- 19. The maximum cap for the service restoration service credit amount shall be a sum mutually agreed upon by MYTV and Access Seeker
- 20. Public information: MYTV must make available, on its publicly accessible website or through interactive self-service portal, access to the Access Seeker on coverage areas and service outages notification in respect of the service in this section.
- 21. Quality of Service (QoS) reporting: On each quarter of the year, MYTV shall notify to the Commission in writing, or any form approved or notified by the Commission on the QoS in respect of the Services set out in this section. For clarification, this report is in addition to the reporting required under subsection 5.3.12 of the MSA.

PART II – NETWORK CO-LOCATION SERVICE

- 1. Network Co-location Service will be provided according to the following terms and conditions.
- 2. MYTV shall allocate space for physical co-location at each POI which is offered to the Access Seeker.

- 3. Forecasts: For the purposes of Paragraph 2 of Section F Part I of this RAO, MYTV shall only request Forecasts where:
 - (a) the maximum period of time covered by Forecasts regarding Network Co-location Service is one (1) year;
 - (b) the minimum intervals or units of time to be used in Forecasts regarding Network Colocation Service is one (1) year; and
 - (c) the maximum frequency to update or to make further Forecasts regarding Network Colocation Service is once a year.
- 4. Acknowledgement of receipt: For the purposes of Paragraph 2.6 of Section F Part II of this RAO, MYTV shall acknowledge receipt of each Order for a Network Co-location Service within two (2) Business Days.
- 5. Time for acceptance or rejection: Subject to any shorter timeframe required under Paragraph 2.15 of Section F Part I of this RAO, MYTV must notify an Access Seeker that an Order for a Network Co-location Service is accepted or rejected within ten (10) Business Days after issuing the Notice of Receipt in respect of the Order.
- 6. Indicative delivery timeframe: For the purposes of Paragraph 2.14 of Section F Part II of this RAO, the indicative delivery timeframe for Network Co-location is twenty (20) Business Days. For clarification, the indicative delivery timeframe in this Paragraph commences from the Notice of Acceptance or confirmation of the Order (as applicable) in accordance with Paragraph 2.9 of Section F Part II of this RAO.
- 7. Billing Cycle: For the purposes of Paragraph 1.10 of Section F Part VI of this RAO, between the Operators, the Billing Cycle for Network Co-location Service shall be one (1) year in advance for the first year and quarterly in advance for subsequent years.
- 8. Inspection: MYTV shall allow nominated employees and/or contractors of a potential Access Seeker to physically inspect network facilities of MYTV during normal business hours provided that the Access Seeker has provided no less than five (5) Business Days' notice of its request to perform a physical inspection and details of its nominees.
- 9. Physical access: Where MYTV provides a Network Co-location Service or for the Access Seeker to perform operations or maintenance activities, MYTV shall allow the Access Seeker, its nominated employees and/or contractors to physically access MYTV's network facilities and the Access Seeker's equipment, and to have physical control over the Access Seeker's equipment located at such network facilities, at equivalent times and in accordance with MYTV's equivalent processes and procedures. MYTV shall not prevent or restrict access to any network facility of site for reasons concerning national or operational security reasons and may only apply reasonable security procedures and processes that comply with the obligations under Paragraph 31(b) of this Section G Part II.
- 10. Nominated personnel: The employees and/or contractors nominated by the Access Seeker under Paragraph 8, 9, 10 and 11 will be reasonable, having regard to:

- (a) the position of each person and the number of persons nominated; and
- (b) the position of each MYTV's personnel and the number of MYTV's personnel to which MYTV provides physical access to such network facilities.
- 11. Escorts: Where MYTV determines that it is necessary to have an escort present when the Access Seeker's nominated employees and/or contractors to physically access MYTV's network facilities, MYTV shall:
 - (a) bear the costs of such escort service;
 - (b) subject to Paragraph 11(d) below, provide immediate physical access to the Access Seeker for emergency maintenance requests, twenty-four (24) hours a day, seven (7) days a week;
 - (c) subject to Paragraph 11(d) below, provide physical access at the time requested by an Access Seeker for planned maintenance requested on the shorter of:
 - (i) two (2) Business Days' notice for manned sites and five (5) Business Days' notice for unmanned sites; and
 - (ii) the period of notice which MYTV requires for itself when providing itself with physical access for planned maintenance;
 - (d) for both planned and emergency maintenance requests at unmanned sites only, MYTV shall have the escort arrive with the shorter of:
 - (i) thirty (30) minutes of time required by the Access Seeker pursuant to Paragraph 11(b) and 11(c) above (as applicable) plus a reasonable window to allow for travel time (which must be estimated in an operations and maintenance manual given to the Access Seeker, or estimated at the time of responding to the Access Seeker's physical access request); and
 - (ii) the period of time MYTV requires for the escorts to arrive for planned and emergency maintenance at unmanned sites.
- 12. Absence of escort: For the purpose of Paragraph 11 above, if an escort does not arrive at MYTV's property within the timeframe specified, the Access Seeker's nominated employees and/or contractors may proceed to enter the property without an escort.
- 13. Site register: The Access Seeker must establish and maintain a register of all persons who visits MYTV's property on the Access Seeker's behalf, which must be made available for inspection by MYTV, upon request.
- 14. Publication of co-location locations and provision of co-location:
 - (a) Subject to Paragraph 31 of this Section G Part II, MYTV shall publish on its publicly accessible website and keep updated a list of the general locations and technically feasible points at which physical co-location is available on and from the date of publication for the following twelve (12) months;

- (b) Subject to Paragraph 16 below, where required due to physical constrains, MYTV shall jointly agree with the Access Seekers as to which Access Seekers should be given the right to physically co-locate at each POI and each network facility and such access shall be granted on a non-discriminatory basis; and
- (c) Where physical co-location cannot be granted to an Access Seeker, the Access Seeker shall be granted either virtual co-location or in-span interconnection as requested by the Access Seeker.
- 15. Deemed Access Provider: If an Access Seeker (referred to in this Paragraph 15 as the "Deemed Access Provider") obtains physical co-location at a POI or a network facility from MYTV, and MYTV is unable to provide virtual co-location or in-span interconnection as required under Paragraph 14 (c) above, it shall be deemed to be an Access Provider for the purposes of this Section G, Part II. The Deemed Access Provider shall be required to permit access to Access Seekers following the same procedures for permitting access as those required to be followed by MYTV. Within two (2) Business Days of reaching a co-location agreement with an Access Seeker, the Deemed Access Provider must notify MYTV of the existence of the agreement and the identity of the Access Seeker, and must ensure that the Access Seeker complies with the relevant co-location obligations contained in this Section G, Part II. The Deemed Access Provider shall be responsible to MYTV for all acts and omissions of any Access Seekers in connection with providing access to Facilities and/or Services under its co-location agreement.
- 16. Lack of space: Subject to Paragraph 17 below, if there are space constraints at a particular location, MYTV shall take reasonable steps to optimise its usage of the space, including through the upgrading of facilities and transferring equipment to an alternative location. After MYTV has used its best efforts to accommodate all Access Seekers, and it is not physically possible for any further Access Seekers to be accommodated, MYTV shall:
 - (a) notify the Commission of the lack of space at the location;
 - (b) provide any supplementary information which may be requested by the Commission (which may include physical inspections by the Commission); and
 - (c) be excused from providing physical co-location at that location unless and until the Commission notifies MYTV that the Commission considers that physical co-location can and must be provided, in which case MYTV shall provide physical co-location as directed by the Commission.
- 17. Reservation of space: MYTV shall not reserve space other than for its own current needs, its future needs (calculated by use of a reasonably projected rate of growth over two (2) years) and the needs of other Access Seekers who are currently occupying or have ordered space from MYTV.
- 18. Allocation of space: MYTV shall allocate space at each location where physical co-location is to be permitted in a non-discriminatory way and will treat other Access Seekers as it treats itself.
- 19. No minimum space requirements: MYTV shall not impose minimum space requirements on an Access Seeker.

- 20. Notice of refusal: If MYTV proposes to refuse, or refuses, a request for physical co-location from an Access Seeker on the basis of current or future needs of MYTV and/or the needs of other Access Seekers who are currently occupying or have ordered additional space from MYTV, it must also notify the Access Seeker and the Commission of:
 - (a) the space currently used by MYTV;
 - (b) the amount of space reserved for MYTV's future need;
 - (c) the space currently occupied by other Access Seekers;
 - (d) the space ordered by other Access Seekers; and
 - (e) the total amount of space potentially available but for the uses set out above.
- 21. Preparatory work by Access Seeker: If preparatory work is necessary for the purposes of allowing the Access Seeker to obtain access to or co-locate at or on MYTV's network facilities, MYTV shall permit the Access Seeker's employees and/or contractors to perform such preparatory work if the Access Seeker satisfies MYTV (acting reasonable and in accordance with the policy referred to in this Paragraph 21) that such employees and/or contractors have the necessary qualifications. MYTV shall publish and make available a policy about the necessary qualifications applicable to employees and/or contractors who will be permitted to perform preparatory work under this Paragraph 21, and such policy to be non-discriminatory in its application to MYTV's personnel and the Access Seeker's employees and/or contractors who perform similar functions.
- 22. Preparatory work by Access Provider: If MYTV agrees to perform preparatory and does so on the basis of an estimated charge (e.g. based on a time and materials basis):
 - (a) MYTV shall not exceed the estimate without providing the Access Seeker with prior written notice that:
 - (i) the estimate will likely be exceeded; and
 - (ii) a further estimate of the charges for the work necessary to complete the preparatory work; and
 - (b) MYTV shall permit the Access Seeker to withdraw the request for preparatory work without penalty if the revised estimate exceeds the original estimate by more than ten percent (10%) of the original estimate.
- 23. Delays: If MYTV agrees to perform preparatory work and it is or likely to be unable to perform such work within the agreed timeframe, MYTV shall:
 - notify the relevant Access Seeker of the delay to a delivery date, together with the reasons for the delay, as soon as practicable after MYTV becomes aware of the possible delay;
 - (b) permit the Access Seeker notified under Paragraph 23(a) above to cancel the preparatory work without penalty if the delay is longer than ten (10) Business Days; and

- (c) compensate the Access Seeker for the costs it has incurred as a result of delay, subject to the Access Seeker using reasonable endeavours to mitigate those costs.
- 24. Utilities and ancillary services: If MYTV has permitted access or physical co-location at a particular location or network facilities, MYTV shall, where the relevant utilities and ancillary services are within MYTV's control, ensure that all necessary utilities and ancillary services are provided to enable the Access Seeker to benefit from such access or co-location to the same extent that MYTV provides to itself, including but not limited to:
 - (a) access to roads;
 - (b) access to land;
 - (c) power, including the provision of back-up power;
 - (d) environmental services (including but not limited to heat, light, ventilation and airconditioning, fire protection);
 - security, taking care to ensure that its agents, representatives or sub-contractors do not damage any equipment, and keeping the location secure and protected from vandalism or theft; and
 - (f) site maintenance.
- 25. Cost: The utility and ancillary costs in respect of the network facilities as contemplated in Paragraph 24 above shall be apportioned (in accordance with fair and equitable principles) between MYTV and all Access Seekers at the relevant location.
- 26. Security caging: MYTV shall not require the use of cages or similar structures to physically segregate co-located equipment, or equipment located at or on network facilities of MYTV.
- 27. Equipment allowance: MYTV shall permit an Access Seeker to locate equipment on or at MYTV's network facilities which is necessary for the purposes of obtaining the benefit of access to the network services and network facilities provided in accordance with this RAO, including but not limited to multi-functional Equipment which may also be used for purposes other than those specified in this Paragraph 27.
- 28. Marking: All Operators shall mark or label their Equipment, wires, cables, batteries and distribution boards in such a manner that they can be easily identified as the property of the relevant Operator. At all times during the Access Seeker's tenancy, it is the responsibility of the Access Seeker to ensure that the marking and labelling is done with reasonable quality.
- 29. Maintenance: MYTV shall permit, and do all things reasonably necessary to allow, an Access Seeker to maintain its Equipment at or on the network facilities to which access has been granted under Paragraph 9 of this Section G Part II of this RAO.
- 30. Extensions: MYTV Shall reasonably permit the Access Seeker, at the Access Seeker's cost, to extend MYTV's network facilities as may reasonably be required to meet Access Seeker's requirements in the circumstances and to the extent technically feasible.

- 31. Security and critical national information infrastructure:
 - (a) MYTV may decline to publish information in connection with particular POI and other locations where Facilities are located, for national or operational security reason, but in such circumstances, MYTV must:
 - (i) promptly provide such information to other Operators on request, subject only to the Operators entering into a confidentiality agreement in accordance with this RAO;
 - (ii) offer to provide, and if the offer is accepted, provide, updated location details to such Operators as POIs and Facilities are withdrawn, introduced and changed; and
 - (iii) provide all such information to the Commission and, on a 6-monthly basis, the locations at which MYTV is offering to supply Network Co-location Service, the locations at which Access Seekers have requested Network Co-location Service and the locations at which MYTV is actively supplying Network Co-location Service.
 - (b) MYTV may establish reasonable security procedures and processes (such as identity checks) to apply to personnel of Access Seekers who will physically access POIs or other locations where Facilities are located. However, such procedures and processes shall:
 - (i) not completely or substantially prohibit an Access Seeker from physically accessing a POI or other relevant location unless MYTV has been directed in writing to do so by the Government (in which case, MYTV shall notify the Commission); and
 - (ii) be no more restrictive or onerous than the procedures and processes that MYTV imposes on its own personnel who physically access the same POIs and locations.

PART III - CHARGES AND CHARGING PRINCIPLES

1. The charges for DTBMS Services will be based on the following formula:

DTBMS Channel = Channel charge (per channel) + Bandwidth charge per Mbps (referring to the following table)

Description	2023	2024	2025
	(RM/year)	(RM/year)	(RM/year)
Channel charge (per channel)	4,876,006	2,952,012	1,028,018
Bandwidth charge (per Mbps)	466,126	639,251	812,377

DTBMS channels rates*:

High Definition Television Channel	2023	2024	2025	
(Mbps)	(RM/year)	(RM/year)	(RM/year)	
4	6,740,510	5,509,016	4,277,526	

Standard Definition Television Channel (Mbps)	2023	2024	2025
	(RM/year)	(RM/year)	(RM/year)
2	5,808,258	4,230,514	2,652,772

Digital Radio Channel (Mbps)	2023	2024	2025
	(RM/year)	(RM/year)	(RM/year)
0.05	4,899,312	2,983,975	1,068,637

^{*}DTBMS channel is inclusive of one (1) audio stream, one (1) subtitling stream and one (1) HbbTV AIT signalling.

[The remainder of this page is intentionally left blank]

2. Additional charges for DTBMS are as follows:

Additional Charges = One-off charges + Recurring charges

One-off charges	RM	
Description		
Activation fee (linear services/per channel):	15,000	
(a) Standard Definition Television Channel		
(b) High Definition Television Channel		
(c) Digital Radio Channel		
Activation fee:		
(a) Surround channel	15,000	
(b) Additional subtitling	3,000	
Activation fee (non-linear services):		
(a) Red button (AIT signalling)	3,000	

Recurring charges for additional DTBMS	RM/year		
Description			
Additional DTBMS sub-service fee:			
Surround Audio (Dolby Digital plus or E-AC3)	89,640		
Additional Audio stream (stereo)	29,880		
Red button (AIT signalling)	3,000		
Additional subtitling	3,000		

Recurring charges for Network Co-location Service	RM	
Description		
(a) Physical Co-location (space)	25 per sq ft	
(b) Electricity with 1 genset back up	0.70 per kW/h	

^{*}The prices for Network Co-location shall be applied for physical co-location for space, sharing of power supply from the same meter, security and maintenance. For the purpose of clarification, other physical co-location services not listed in the table above are negotiated charges.

[The remainder of this page is intentionally left blank]

ANNEXURE 1: DISPUTE RESOLUTION PROCEDURES

1. Introduction

- 1.1 Subject to Paragraph 1.1(b) of this Annexure, MYTV and an Access Seeker shall adopt and comply with these Dispute Resolution Procedures in relation to any dispute which may arise between the Parties in relation to or in connection with the supply of Services to which this RAO applies ("Dispute"). The dispute resolution mechanisms are as follows:
 - (a) Interconnect steering group; and
 - (b) subject to specific resolution of disputes, being:
 - i. Technical Disputes (which must follow the procedure set out in Paragraph 4 of this Annexure if they cannot be resolved through the application of the general dispute resolution provisions in Paragraphs 2 and 3 of this Annexure);
 - ii. Billing Disputes, which must follow the procedures set out in Paragraph 5 of this Annexure; or
 - iii. any other types of disputes, which, if cannot be resolved through the application of the general dispute resolution provisions in Paragraphs 2 and 3 of this Annexure, must be referred to the Commission for resolution.
- 1.2 A Dispute shall first be attempted to be resolved by negotiation between the Parties. If the Parties to the Disputes cannot or otherwise fail to reach an agreement, the Parties shall always be entitled to seek resolution of the Dispute by the Commission in accordance with Section 151 of the Act, and the Commission will decide the dispute if it is satisfied that:
 - (a) the Parties will not reach agreement, or will not reach agreement in a reasonable time;
 - (b) the notification of the Dispute is not trivial, frivolous or vexatious; and
 - (c) the resolution of the Dispute would promote the objects in the Act.

MYTV shall not prevent the Access Seeker from notifying a Dispute to the Commission in accordance with the Act.

1.3 For clarification, unless stated otherwise, all references to sections and paragraphs in this Annexure are references to sections and paragraphs of this Annexure.

2. General

2.1 An Operator may not commence court proceedings relating to that a Dispute which is the subject of these Dispute Resolution Procedures until it has complied with each applicable process in these Dispute Resolution Procedures, other than an application for urgent interlocutory relief. Nothing in this paragraph shall be construed as ousting the jurisdiction of any court.

- 2.2 Both Parties to a Dispute shall ensure that their representative acting in relation to a Dispute are of sufficient seniority and have authority to settle a Dispute on their behalf. At the commencement of the Dispute Resolution Procedures, each party must notify the other party of the scope of the authority of each of their representatives. If, in the course of the Dispute Resolution Procedures, it is identified that the matters to be resolved are outside the initial term of reference for which authority was given to a representative, a party may require that those matters be referred to more senior officers of that Operator who have authority to settle those matters.
- 2.3 During a Dispute and any dispute resolution process invoked in accordance with this Annexure, the Parties must continue to fulfil their obligations under the Access Agreement between them.
- 2.4 Subject to Paragraph 2.5 of this Annexure, the Parties to a Dispute shall exchange information of a type described in this RAO during the course of, and to facilitate, resolution of the Dispute.
- 2.5 Confidential Information of a party which is disclosed, and any other oral or written submissions made by a party or a party's representatives during the course of any dispute resolution process will be subject to the confidentiality restrictions in relevant confidentiality provisions contained in the Confidentiality and Non-Disclosure Agreement prepared in accordance with Annexure 2 in this RAO.
- 2.6 A party must not use information obtained under Paragraph 2.4 of this Annexure or described in Paragraph 2.5 above for any purpose other than to resolve the Dispute.
- 2.7 Subject to Chapter 7 of Part V of the Act, an arbitrator of a Dispute (including a Technical Expert or the Commission, in accordance with this Annexure) may decide not to determine the Dispute if the arbitrator considers that the Dispute is trivial, frivolous or vexatious, or if there is insufficient evidence before the arbitrator to determine the Dispute.
- 2.8 The costs of the arbitration are to be shared equally between the Parties, unless the arbitrator of the Dispute has decided not to determine the Dispute in accordance with Paragraph 2.7 above. If an arbitrator decides not to determine the Dispute, the party that initiated the Dispute must pay the other party's costs.

3. Interconnect Steering Group

- 3.1 In the first instance the Parties should attempt to resolve the Dispute between themselves. Either party may give written notice ('Notice') to the other party ('Receiving Party') stating its intention to form, within ten (10) Business Days, an Interconnect Steering Group ('ISG') and outline the details of the Dispute.
- 3.2 The Parties shall form the ISG within ten (10) Business Days, to fulfil the requirements of Paragraph 3.1 above. The ISG shall comprise of representatives of the Parties, and be headed by a person who holds a position that is at least equivalent to the Chief Officer or Executive Vice President of MYTV..
- 3.3 The parties shall provide for:

- (a) subject areas to be dealt with by ISG;
- (b) equal representation by the Parties;
- (c) chairmanship and administrative functions of the ISGto be shared equally; and
- (d) formal notification procedures to the ISG.
- 3.4 The Parties shall use reasonable endeavours to attempt to settle the Dispute in the ISGfor a period of no longer than thirty (30) Business Days unless otherwise agreed by the Parties, subject always to a party's right to seek urgent interlocutory relief.
- 3.5 In the even that the Parties cannot resolve the Dispute between themselves within the time specified in Paragraph 3.4 above, or after any agreed time extension has expired, either party may notify the other party that it wishes to refer the issue to:
 - (a) to the extent the issues in dispute are technical in nature, to a Technical Expert (in accordance with Paragraph 4 of this Annexure); or
 - (b) to the Commission for final arbitration.
- 3.6 The ISG to which an issue has been raised will meet within ten (10) Business Days of the receipt by the Receiving Party of the Notice under Paragraph 3.1 of this Annexure. If ISG fails to meet or has not been formed within ten (10) Business Days of the receipt by the Receiving Party of the Notice, either Party may refer the Dispute:
 - (a) to the extent the issues in dispute are technical in nature, to a Technical Expert (in accordance with Paragraph 4 of this Annexure); or
 - (b) to the Commission for final arbitration.

4. Use of a Technical Expert

- 4.1 A Dispute will only be referred to a Technical Expert if the provisions of this RAO have been complied with.
- 4.2 Once a Dispute is referred to a Technical Expert, it may not be referred back ISG.
- 4.3 The person to whom a technical dispute may be referred under this Paragraph 4:
 - (a) will be an expert appointed by agreement of the Parties or, if the Parties cannot agree, by the Commission;
 - (b) will have the appropriate qualifications and experience to arbitrate the dispute, including knowledge of the communications industry;
 - (c) need not be a Malaysian citizen or resident; and

(d) will not be an officer, director, or employee of a communications company or otherwise have a potential for conflict of interest,

("Technical Expert").

- 4.4 If the Parties fail to appoint a Technical Expert within ten (10) Business Days of the need to refer a Dispute to a Technical Expert, a Technical Expert will be appointed by the Commission.
- 4.5 When relying on the services of a Technical Expert, the following dispute resolution procedure will apply to the Technical Expert:
 - (a) the Parties will present written submissions to the Technical Expert and each other within fifteen (15) Business Days of the appointment of the Technical Expert; and
 - (b) each party may respond to the other party's submission in writing within fifteen (15) Business Days from the date of the other party's submission. No further submissions in reply shall be made except with the Technical Expert's approval.
- 4.6 At the request of either party and subject to the parties agreeing, or the Technical Expert deciding within five (5) Business Days of the last written submission, that the arbitration by the Technical Expert should be by documents only, a Technical Expert hearing will be held within fifteen (15) Business Days of the last written submission.
- 4.7 Should a Technical Expert hearing be held, each party will have the opportunity of making an oral submission in addition to the written submissions submitted in Paragraph 4.5 and 4.6 above. This process will be conducted in private.
- 4.8 The procedure for hearing technical disputes will be determined by the Technical Expert (including number and duration of oral submissions by the Parties) but in any case, the Technical Expert's hearing will last no longer than three (3) Business Days.
- 4.9 The Technical Expert will not have the power to appoint any other experts.
- 4.10 The Technical Expert will deliver his or her award within fifteen (15) Business Days of the hearing or of the last written submission where the arbitration is by documents only.
- 4.11 Every Dispute referred to a Technical Expert will be considered separately so that time limits for each dispute are complied with.
- 4.12 The Technical Expert's decision will be binding on the Parties (in the absence of manifest error of fact or law).

5. Billing Dispute resolution

5.1 As outlined in the billing provisions of this RAO at Part VI, a party ("Invoicing Party") shall provide to the other party ("Invoiced Party") an Invoice in writing, or in such electronic form as may be agreed from time to time, for amounts due in respect of the supply of Services during such Billing Period.

- 5.2 An Invoicing Party shall allow an Invoiced Party to dispute an Invoice prepared by the Invoicing Party if the Invoiced Party notifies the Invoicing Party within thirty (30) Business Days after the date of receipt of such invoice, provided that, in any case specified above, the Invoiced Party's Billing Dispute Notice specifies the information in accordance with Paragraph 5.4 of this Annexure.
- 5.3 A Billing Dispute may only arise where the Invoiced Party has reasonable grounds to believe that an error has arisen from one of the following circumstances:
 - (a) the Invoicing Party's Billing System is, or has been, defective or inaccurate in respect of the recording of the calls which are the subject of the Dispute;
 - (b) there is, or has been, a discrepancy between the Invoice in dispute and the records generated by the Invoiced Party's Billing System;
 - (c) there is, or has been, a fraud perpetrated by the Invoicing Party; or
 - (d) the Invoicing Party has made some other error in respect of the calculation of the charges which are the subject of the Billing Dispute.
- 5.4 A Billing Dispute Notice given under this Paragraph 5 must specify:
 - (a) the reasons for which the Invoiced Party disputes the Invoice is disputed;
 - (b) the amount in dispute;
 - (c) details required to identify the relevant Invoice and charges in dispute including:
 - (i) the account number;
 - (ii) the Invoice reference number;
 - (iii) the Invoice date;
 - (iv) the Invoice amount; and
 - (v) billing verification information; and
 - (d) evidence in the form of a report, indicating the relevant traffic data which is in dispute.
- 5.5 The Invoiced Party may withhold payment of amounts disputed in good faith in accordance with Paragraph 1.15 of Part VI of this RAO. If the Billing Dispute is resolved against the Invoiced Party, that Invoiced Party shall be required to pay interest at the rate specified in Paragraph 1.19 of Part VI of this RAO on the amount payable.
- 5.6 Where the Invoiced Party has paid an amount and subsequently notifies the Invoicing Party of a Billing Dispute in relation to that amount within the Billing Dispute Notification Period, the Invoicing Party is not obliged to refund any or all of that amount until the Billing Dispute is resolved in respect of that amount. Once the Billing Dispute is resolved, if the Invoicing Party is

obliged to refund an amount to the Invoiced Party, interest will be payable on the refunded amount at the rate specified in Paragraph 1.19 of Part VI of this RAO. In such circumstances, interest will be payable from the date the Invoiced Party paid the disputed amount to the date of the refund by the Invoicing Party.

- 5.7 The parties agree to use their reasonable endeavours to promptly resolve any Billing Dispute notified under this Paragraph 5.
- 5.8 If the parties are unable to resolve any Billing Dispute within one (1) month (or such other period as the parties may agree) from the date on which the Billing Dispute Notice is received, either party may seek the consent of the other party to extend the period for resolution of the Billing Dispute stating the exceptional reasons for such extension. The other party is, however, under no obligation to agree to such extension.
- 5.9 To the extent that a Billing Dispute notified under this Paragraph involves a Billing Dispute with an international correspondent of the Invoicing Party, the Dispute Resolution Procedures shall be suspended for a reasonable period of time pending resolution of the Billing Dispute with that international correspondent. As a general rule, the period of suspension will not exceed four (4) months. However, the parties shall recognise that some Billing Disputes with international correspondents may take longer to resolve, in which case the Invoicing Party must promptly inform the Invoiced Party of the likely period required for resolution.
- 5.10 Once the negotiation period under Paragraph 5.8 of this Annexure (including any extension agreed) and any suspension period under Paragraph 5.9 of this Annexure have expired, the Billing Dispute may be referred by the Invoiced Party to the procedure described in Paragraph 5.11 of this Annexure ("Billing Dispute Escalation Procedure").
- 5.11 The Invoiced Party may refer a Billing Dispute to the Billing Dispute Escalation Procedure under this Paragraph 5.11 by notifying the Invoicing Party's Billing Representative. Both parties shall then appoint a designated representative that who has authority to settle the Billing Dispute, and who is at a higher level of management than the persons with direct responsibility for administration of this RAO. The designated representatives shall meet as often as they reasonably deem necessary in Order to discuss the Billing Dispute and negotiate in good faith in an effort to resolve such Billing Dispute. The specific format for such discussions will be left to the discretion of the designated representatives, however all reasonable requests for relevant information made by one party to the other party shall be honoured.
- 5.12 Once any Billing Dispute has been resolved to the parties' satisfaction, any sum to be paid or repaid shall be paid by the relevant party within ten (10) Business Days from the date of resolution of the Billing Dispute.
- 5.13 Although it shall be the good faith intention of the parties to use the above Billing Dispute Resolution Procedures to the fullest extent to try to solve Billing Disputes, nothing in this Annexure shall prevent either party from pursuing any other remedy in law or equity that may be available to them if a Billing Dispute cannot be resolved to their satisfaction.
- 5.14 A party may request a joint investigation of Invoice discrepancies after that party has conducted a comprehensive internal investigation, including an examination of its own Billing System. Prior to commencement of the joint investigation, the Parties must agree on the

terms of the joint investigation, including:

- (a) the scope of the joint investigation;
- (b) how the joint investigation will be conducted; and
- (c) the date by which the joint investigation must be concluded.

The joint investigation may include the generation of test calls to the other party's Network.

- 5.15 Enquiries relating to billing, collecting and settlement arrangements or in relation to Network and operation issues may be directed to the Billing Representatives nominated by each party.
- 5.16 Either party may at any time nominate another Billing Representative, provided that ten (10) Business Days prior notification of such appointment is given.
- 5.17 If the Billing Dispute Escalation Procedure has been exhausted, either party may refer to the Billing Dispute to the Commission for resolution under Chapter 7 of Part V of the Act.

[The remainder of this page is intentionally left blank]

ANNEXURE 2: CONFIDENTIALITY AND NON-DISCLOSURE AGREEMENT

This Confidentiality and Non-Disclosure Agreement ("Agreement") is made onday of("Effective Date") between:
MYTV Broadcasting Sdn. Bhd. (Company No. 201001013318 / 897549-X), a corporation whose principal place of business is at 2271, Jalan Usahawan 2, Cyber 6, 63000 Cyberjaya, Selangor ("MYTV") of one part;
and
Company Name (Company No. XXX-X), a corporation whose place of business is at("The Company") of the other part.
(MYTV and The Company shall collectively be referred to as the "Parties" and individually as the "Party")
WHEREAS:

- (A) The Parties wish to disclose certain Confidential Information (as defined in Clause 1) to each other for the purpose of conducting a consumer survey to collect data and information from the recipients of MYTV's Set Top Boxes ("STBs") as to understand and measure their receptiveness and feedback on the usage of STBs and related matters ("Purpose"), and this will necessitate the sharing of confidential and/or proprietary information.
- (B) Each of the Parties wish to protect its position in relation to the Confidential Information so disclosed in relation to the Purpose, whether such Confidential Information is disclosed visually, orally, or in the form of photographs, drawings, pictorial, electronic or in other written material, machine readable data and human readable form, software or hardware together with any analysis, compilations, studies, presentation or other documents prepared by either of the Party in relation to the Confidential Information or derived from or pursuant to discussions with any of its affiliates or the officers, employees, agents, advisers, consultants or representatives thereof in relation to the Confidential Information.

NOW THEREFORE, in consideration of these premises, the Parties hereto agree as follows:

- 1. In this Agreement unless the context otherwise require the following expressions shall have the meanings respectively set out below:
 - "Affiliate" of a Party means in relation to that Party (i) any entity under the Control of such Party; (ii) any entity Controlling such Party; and (iii) any other entity under the Control of a Controlling entity under the preceding Paragraph (ii). "Control" or "Controlling" means the owning of fifty percent (50%) or more of the equity interest or any other type of ownership by one Party in another incorporated entity.

"Confidential Information" shall include but not be limited to:

a) information, knowledge or data relating to sales, commercial matters, products and their price, financial or marketing plans and information, or information of an intellectual or non-technical form including intellectual property or technical nature such as technical data, sketches, models, inventions, know-how, processes, apparatus, equipment, business plans, financial simulations, information concerning employees, research, experimental work, development, design details and

specifications, engineering, financial information, procurement requirements, purchasing, manufacturing, customer lists, business forecasts, sales and merchandising, drawings, photographs, cinematographic works, charts, diagrams, specifications, standards, manuals, reports, formulae, algorithms, processes and trade secrets derived from the Confidential Information or other information in which the Disclosing Party has a business, proprietary or ownership interest or legal duty to protect and which is not yet in the public domain including any information which is either clearly marked as 'Confidential' or unmarked, on which the Disclosing Party has included information which is confidential, or information relating in any way to the Purpose;

- information of whatever nature relating to the business of the Company obtained by observation during visits (if any) to its premises; and
- c) information relating to "Personal Data" in accordance with Personal Data Protection Act 2010.

"Disclosing Party" means a Party (through any of its Affiliates or the officers, employees, agents, advisers, consultants or representatives thereof or of the Party), who provides, discloses or supplies Confidential Information to the Receiving Party.

"Receiving Party" means a Party (through any of its Affiliates or the officers, employees, agents, advisers, consultants or representatives thereof or of that Party), who receives Confidential Information from the Disclosing Party.

- 2. All Confidential Information to be disclosed orally shall be made in a positive manner in an appropriate forum prior to its disclosure and must be confirmed in writing within fourteen (14) days of the oral disclosure by the Disclosing Party to the Receiving Party during which time such Confidential Information disclosed orally shall be similarly protected.
- 3. In respect of all Confidential Information disclosed, the Receiving Party undertakes, subject to Clause 4 below:
 - i. to keep such Confidential Information in strict confidence, and not to use the Confidential Information otherwise than for the purposes of the Purpose;
 - ii. only to disclose the Confidential Information to such of its or its Affiliates' officers, employees, agents, advisers, consultants or representatives as on a genuinely "need to know" basis only for the purposes of this Agreement, and then only on the understanding that they agree to be similarly bound by the provisions of this Agreement. The Receiving Party shall procure such of its officers, employees, agents, advisers, consultants or representatives to comply with the terms hereof. In any event, the Receiving Party shall be responsible for any breach of the terms of this Agreement by any of its officers, employees, agents, advisers, consultants or representatives and shall take all measures to restrain such officers, employees, agents, advisers, consultants or representatives from prohibited or unauthorised disclosure or use of the Confidential Information;
 - iii. not to disclose the Confidential Information to any person or entity not explicitly named in this Agreement, particulars of the Parties that have exchanged the Confidential Information, discussions and negotiations which are taking place concerning the Purpose between the Parties, any of the terms, conditions or other facts with respect to the Purpose including the status thereof, without prior written consent of the Disclosing Party;
 - iv. to maintain and protect the Confidential Information so disclosed with the same degree of care and control as used to keep confidential its own Confidential Information. The Receiving Party shall restrict access to the Confidential Information so as to prevent its unauthorised use, disclosure or reproduction and shall maintain reasonable procedures to prevent loss of any

Confidential Information;

- v. not to use or cause to be used any Confidential Information in the course or pursuit of any other business relationship or otherwise with any person or entity which is acting or may act in business competition with the Disclosing Party or to secure a competitive advantage over the Disclosing Party or cause, suffer or permit to be done any of the foregoing acts; and
- vi. to return to the Disclosing Party on demand or expiry or termination of this Agreement all Confidential Information including all copies thereof, and to destroy or to certify its destruction all notes and any other written reports of documents which may have been made by the Receiving Party and which contain any part of the Confidential Information, except as authorised in writing by the Disclosing Party.
- 4. The obligations in Clause 3 above shall not apply to any Confidential Information which:
 - i. the Receiving Party can show that the Confidential Information was already in or came into the public domain prior to the date of this Agreement or thereafter becomes publicly available without any violation of this Agreement or otherwise than by a breach of this Agreement;
 - ii. the Receiving Party can show that the Confidential Information was already in its possession free of any restriction as to non-disclosure, prior to, at or subsequent to the time of receipt from the Disclosing Party or becomes available to a Party from a person, other than the other Party and its representatives, who is not, to the best of such Party's knowledge subject to any legally binding obligation to keep such information confidential;
 - iii. the Receiving Party can prove that the Confidential Information has been independently developed or generated independently without access to or use of or reference to any Confidential Information disclosed under this Agreement;
 - iv. is disclosed by the Receiving Party to satisfy a requirement of, or demand by, a competent court of law or governmental or regulatory body made pursuant to law or to satisfy the requirements of any stock exchange upon which shares of the parties are listed, provided, however, to the extent permissible under the laws and regulations, that the Receiving Party shall notify and consult the Disclosing Party as to the form, nature and purpose of the disclosure to enable it to seek a protective order or otherwise prevent such disclosure.
- 5. This Agreement shall not be construed as granting expressly or by implication at any time, any transfer, assignment, license on any other rights in respect of any licence, patent, copyright, trademarks or other intellectual property right in force and belonging to the Disclosing Party, which rights shall remain vested in, and the absolute property of the Disclosing Party.
- 6. The property in all Confidential Information disclosed under this Agreement including all copies thereof shall, subject to any right of any other owner, rest with the Disclosing Party. Should either Party be subject to re-organisation, merger, takeover or the like, its successor in law shall be bound by the conditions hereof as if they were the original party hereto.
- 7. In the event that the Receiving Party is obligated to disclose any Confidential Information as a result of any applicable law, rule or regulation or a court order or pursuant to governmental action, the Receiving Party shall immediately inform (or where Immediate notice is not possible, as soon as practicable thereafter) the Disclosing Party so that the Disclosing Party is given an opportunity to object to such disclosure. Should any such objection by the Disclosing Party be unsuccessful, the Receiving Party so obligated to disclose Confidential Information may disclose only such Confidential Information to the extent required by the relevant court order or governmental action.
- 8. The Receiving Party agrees that, in the event of a breach or threatened breach of the terms of this

Agreement, the Disclosing Party shall be entitled to an injunction in addition to and not in lieu of any other legal or equitable relief including money damages. The Receiving Party acknowledges that the Confidential Information is valuable and unique and that disclosure will result in irreparable injury to the Disclosing Party.

- 9. Neither Party shall advertise or publicly announce that it has entered into this Agreement without the prior written consent of the other Party, such consent not to be unreasonably withheld, nor shall either Party make any public announcement regarding the Purpose or use the other Party's name for promotional or marketing purposes.
- 10. If any provision of this Agreement is held to be invalid, illegal or unenforceable then such provision shall be automatically modified to the extent necessary to make it valid, legal and enforceable whilst preserving the intent of the Parties and all other provisions of this Agreement shall be regarded as fully valid and enforceable unless otherwise proved.
- 11. The Disclosing Party makes no warranties regarding the accuracy of such Confidential Information. The Disclosing Party accepts no responsibility for any expenses, losses, or action incurred or undertaken by the Receiving Party as a result of the receipt of the Confidential Information. It is further understood by the Receiving Party that the Disclosing Party does not warrant or represent that it will introduce any product or service to which the Confidential Information disclosed herein is related.
- 12. This Agreement shall be effective upon its execution, and shall, unless terminated by either Party providing one (1) month's written notice of termination to the other Party, continue for a period of two (2) years from the date of this Agreement ("Term"), provided however that the obligations undertaken herein with respect to Confidential Information received prior to the termination of this Agreement shall survive and continue for a period of two (2) years after any expiration or termination of this Agreement or upon the execution of the subsequence/definitive agreement to the Purpose, whichever is the earlier, which shall incorporate confidentiality provisions on similar form and content as herein contained.
- 13. The Parties hereto acknowledge that a breach of any of the promises or agreements contained herein will result in irreparable and continuing damage to either Party for which there will be no adequate remedy at law, and the affected Party shall be entitled to injunctive relief and/or decree for specific performance, and such other relief as may be proper (including monetary damages if appropriate).
- 14. All notices, demands or other communications under this Agreement must be given or made in writing, and must be delivered personally, by overnight courier, by A.R. registered mail or sent by fax or email transmission, with electronic confirmation of receipt, as follows:

If to:

MYTV Broadcasting Sdn. Bhd. The Company

2271, Jalan Usahawan 2,

Cyber 6,

63000 Cyberjaya,

Selangor

Telephone: 03-8313 8606 Telephone : [the company's phone number

1

Attention: Chief Operating Officer Attention : [the company's CEO/COO

1

or at such address or to such fax number as may be designated by notice from such Party.

15. This Agreement and the rights and obligations hereunder are personal to the parties and may not be assigned or otherwise transferred, in whole or in part, without prior written consent of both parties.

- 16. Each Party shall bear its own costs and expenses incurred in connection with the Purpose and this Agreement.
- 17. The construction, validity and performance of this Agreement shall be governed by Malaysian law, and the Parties hereto submit to the exclusive jurisdiction of the Malaysian courts.
- 18. Each Party's Confidential Information shall be held in strict confidence by the other Party, using no lesser security measures and degree of care as it uses to protect its own Confidential Information. In any event, the security measures and the degree of care it uses shall, as a minimum, comply with the standards imposed by the applicable laws including the Personal Data Protection Act 2010. The Receiving Party shall further ensure that the Confidential Information is secured from unauthorised access from internal and external parties and that all Confidential Information used, stored and/or processed shall be free from virus, malware or other malicious codes.

IN WITNESS WHEREOF the Parties hereto have caused this Agreement to be executed by their duly authorised officers the day and date first before written.

Signed for and on behalf of MYTV BROADCASTING SDN. BHD. (Company No. 201001013318 / 897549-X) in the presence of: Name: Name: Designation: Designation: NRIC No: Signed for and on behalf of (Company No.) in the presence of: Name: Name:

Designation:

NRIC No.:

Designation:

NRIC No.:

ANNEXURE 3: APPLICATION FORM

Digital Terrestrial Television Malaysia Access Seeker Application Form

As iterated in the RAO document, an Access Seeker that wishes to obtain access to any Service from MYTV Broadcasting Sdn. Bhd. must submit an Access Request to MYTV. The Access Request shall contain the following information;

- (a) the name and contact details of the Access Seeker;
- (b) the Services in respect of which Access is sought;
- (c) preliminary information regarding the scale and scope of Services that the Access Seeker expects to acquire from MYTV pursuant to the Access Request;
- (d) relevant technical information relating to the interface Standards of the Access Seeker;
- (e) creditworthiness information in accordance with MYTV's requirements;
- (f) assessed security (or, if applicable, confirmation of security provided) in accordance with MYTV's requirements;
- (g) insurance information in accordance with MYTV's requirement;
- (h) bank guarantee information for security sum (12 months of service fee per channel); and
- such other information as MYTV may reasonably request for the sole purpose of providing access to the requested Services.

Declaration

We hereby declare that all information provided in this application is true and agree that MYTV Broadcasting Sdn. Bhd. reserves the right to review this application at its sole discretion.

We confirm that we are not in any breach of any confidentiality in providing the information herein. We also have no objection for MYTV Broadcasting Sdn. Bhd. to cross-reference to a third party, visit or audit our premises and facilities. We will extend our assistance towards this exercise. We also fully understand that nothing in this proposal is intended to constitute an agency, partnership or joint venture between the parties.

This registration shall terminate, without notice, (i) upon the institution by or against either party of insolvency, receivership or bankruptcy proceedings or any other proceedings for the settlement of either party's debts, (ii) upon either party making an assignment for the benefit of creditors, or (iii) upon either party's dissolution or ceasing to do business. The requestor shall ensure that there are no pending or overtly threatened actions, claims, orders, decrees, investigations, suits or proceedings by or before any governmental authority, arbitrator, court or administrative agency which would have a Material Adverse Effect, which might result in any impairment of the right or ability of the requestor to enter into or perform his obligations under this Agreement.

Personal Data Protection Act 2010

- (i) Personal Data of the Customer and/or third party
 - a) Subject to the terms and conditions of this Agreement, where the "interested consultant/vendor/contractor" receives or process any personal data (as defined by Personal Data Protection Act 2010) ("PDPA") from any party on behalf of MYTV Broadcasting Sdn. Bhd. arising from the performance of this Agreement, it shall ensure that it fully complies with the provisions of the PDPA and only deals with the data to fulfill its obligations under the Agreement.
 - b) Without prejudice to any other "interested consultant/vendor/contractor" obligation to indemnify MYTV Broadcasting Sdn. Bhd. under this Agreement, the "interested consultant/vendor/contractor" shall indemnify MYTV Broadcasting Sdn. Bhd. for any breach of the PDPA which renders the latter for any costs, claims or expenses.
 - c) In fulfillment of its obligations under the PDPA, the "interested consultant/vendor/contractor" shall have a system in place to ensure:
 - i) full compliance with the PDPA; in particular the principles of the PDPA which deals with the security of personal data;
 - ii) the reliability of all its employees who may be involved in receiving and processing the personal data
 - d) The "Interested consultant/vendor/contractor" shall take reasonable steps to ensure that all of its partners, shareholders, directors, employees, contractors and agents comply with this Clause where they are receiving and processing any personal data on behalf of MYTV Broadcasting Sdn. Bhd. during its performance of this Agreement.

- e) The "Interested consultant/vendor/contractor" shall allow MYTV Broadcasting Sdn. Bhd. reasonable access to such information as is necessary to ensure that it is complying with the above provisions and the PDPA as a whole.
- f) Immediately following termination or expiry of this Agreement, MYTV Broadcasting Sdn. Bhd. partners are obliged to comply and perform its obligations under this Agreement in respect of all personal data in its possession or control.
- (ii) "Interested Broadcaster/Vendor/Contractor" Personal Data
 - a) The "Interested broadcaster/vendor/contractor" agrees and understands that by providing its personal data and entering into this Agreement, the "interested broadcaster /vendor/contractor" acknowledges that the "interested broadcaster /vendor/contractor" has given MYTV Broadcasting Sdn. Bhd. its consent to collect, record, hold, store, use and disclose its personal information for purposes which are necessary or related to this Agreement by MYTV Broadcasting Sdn. Bhd. according to the PDPA and MYTV Broadcasting Sdn. Bhd.'s Privacy Statement.

TYPE OF INCORP	ORATION		
Sole Proprietor			
Private Limited			
Limited			
Date of Incorpor	ation:		
Place of Incorpor	ration:		
Company Registr	ation No.:		
Year of Registrat	ion:		
FINANCIAL INFO	RMATION		
CAPITAL STRUCT	URE		
Authorised Capit	al		
Paid-Up Capital			
Working Capital			
FINANCIAL FACII	LITIES		
Туре		Institution	Value (RM)
BANK ACCOUNT			
Bank		Branch	Account No.

CORPORATE STRUCTURE

BOARD OF DIRECTORS

NAME	PASSPORT / I.C NO.	NATIONALITY	DESIGNATION	% OF SHAREHOLDING

NUMBER OF PERMANENT STAFF

MANAGEMENT	EXE	CUTIVE	NON-	EXECUTIVE	TOTAL
	Technical	Non-Technical	Technical	Non-Technical	

CERTIFICATION

Please attach copies of the following documents:

- I. Company Profile
- II. Form 9 Registration with Suruhanjaya Syarikat Malaysia
- III. Form 24 Return of Allotment of Shares
- IV. Form 49 Return giving particulars in register of Directors, Managers and Secretaries and changes of particulars

Digital Terrestrial Television Malaysia Service Summary

Company Name			Company					
				Reg. No:				
Company Address				CASP				
				License				
				Number:				
Account Ma	nager							
				Logical				
Service	Name			Channel				
			Number					
			(LCN):					
	HD			Number of				
Video				audio pair				
Service	SD			Number of				
				Subtitles				
(Genre			Radio				
				Service				
Service Activ	ation			Interactive				
	Date:			Signalling	YES	NO		
Requested by:				Received by (Operator):				
Name: Designation: Company: Date:			Name: Designation: Company: Date:					

ANNEXURE 4: LIST OF TRANSMITTERS SITES

NO.	SITE NAME	STATE	REGION	LOCATION	COVERAGE AREA	AFFECTED POPULATION	CATEGORY (REFER TO TABLE B)
1	DTH Uplink	N/A	N/A	N/A	Satellite (DTH) coverage is available for nationwide (Malaysia)	347,609*	(A)
2	Menara KL	KL	CTL	KL	Kuala Selangor, Bestari Jaya, Puncak Alam, Sungai Buloh, Shah Alam, Puchong, Subang, Klang, Banting, Tanjung Sepat, Sepang, Kajang, Kuala Lumpur, Batu Caves, Gombak, Selayang, Ampang, Petaling Jaya, Rawang	6,353,319	(A)
3	Gunung Jerai	KDH	NORTH	Gurun	Kuala Perlis, Simpang Ampat, selatan Kangar, selatan Arau, Sanglang, Ayer Hitam, Kodiang, Jitra, Pendang, Bandar Baharu, Kota Setar, Pokok Sena, Kuala Muda, Yan, Kubang Pasu, Kulim, Padang Terap, Pendang, Gurun, Sungai Petani, Padang Serai, Kuala Ketil, Kepala Batas, Butterworth	2,249,149	(A)

NO.	SITE NAME	STATE	REGION	LOCATION	COVERAGE AREA	AFFECTED POPULATION	CATEGORY (REFER TO TABLE B)
4	Bukit Sungai Besi	SEL	CTL	Cheras	Tanjung Sepat, Sepang, Banting, Beranang, Semenyih, Bangi, Kajang, Cheras, Putrajaya, Cyberjaya, Kuala Lumpur, Petaling Jaya, Puchong, Subang, Shah Alam, Klang, Ampang, Gombak, Batu Caves, Selayang, Subang, Sungai Buluh, Nilai, Pajam, Mantin, Labu, Sebahagian Seremban	2,090,082	(A)
5	Gunung Ledang	JHR	SOUTH	Tangkak	Batu Pahat, Parit Raja Yong Peng, Kluang, Pagoh, Muar, Segamat, Labis, Tangkak, Melaka, Gemencheh, Gemas	2,006,420	(A)
6	Gunung Ulu Kali	PHG	EAST	Genting Highlands	Sabak Bernam, Sungai Besar, Sekinchan, Tanjung Karang, Kuala Selangor, Bestari Jaya, Puncak Alam, Sungai Buloh, Shah Alam, Puchong, Subang, Klang, Banting, Kajang, Kuala Lumpur, Petaling Jaya, Rawang, Bukit Beruntung, Serendah, Batang Kali, Kuala Kubur baru, Tanjung Malim, Bentong, Bera, Temerloh, Mentakab, Teriang, Raub, sebahagian Jengka, Kuala Krau	887,749	(A)

NO.	SITE NAME	STATE	REGION	LOCATION	COVERAGE AREA	AFFECTED POPULATION	CATEGORY (REFER TO TABLE B)
7	Gunung Kledang	PRK	NORTH	Ipoh	Kuala Kangsar, Sungai Siput, Chemor, Ulu Kinta, Ipoh, Lahat, Batu Gajah, Kampar, Tapah, Bidor, Sungkai, Slim River, Langkap, Teluk Intan, Kampung Gajah, Sitiawan, Bota, Parit, Manong, Beruas, Pantai Remis	1,698,258	(A)
8	Bukit Karatong	SBH	SBH	Kota Kinabalu	Membakut, Kota Kinabalu, Penampang, Putatan, Tuaran, Tamparuli, Papar, Kinarut, Kimanis, Kuala Penyu	1,097,770	(A)
9	Gunung Pulai	JHR	SOUTH	Kulai	Senggarang, Simpang Renggam, Benut, Bandar Tenggara, Nusajaya, Gelang Patah, Johor Bahru, Kota Tinggi, Kulai, Senai, Pontian, Kukup, Ulu Tiram	1,259,274	(A)

NO.	SITE NAME	STATE	REGION	LOCATION	COVERAGE AREA	AFFECTED POPULATION	CATEGORY (REFER TO TABLE B)
10	Gunung Serapi	SWK	SWK	Kuching	Asajaya, Bau, Kuching, Lundu, Padawan, Kota Samarahan, Sematan, Sebangan, Siburan	952,664	(A)
11	Bukit Bakar	KEL	EAST	Machang	Kuala Krai, Machang, Pasir Putih, Tanah Merah, Ketereh, Selising, Bachok, Melor, Rantau Panjang, Pasir Mas	797,235	(A)
12	Bukit Pelindung	PHG	EAST	Kuantan	Cherating, Gebeng, Balok, Bukit Goh, Pekan, Kuantan	646,241	(A)

NO.	SITE NAME	STATE	REGION	LOCATION	COVERAGE AREA	AFFECTED POPULATION	CATEGORY (REFER TO TABLE B)
13	Bukit Besar	TRG	EAST	Kuala Terengganu	Kuala Berang, Bukit Payong, Merang, Kuala Nerus, Kuala Terengganu, Marang, Merchang	556,502	(A)
14	Bukit Trig	SBH	SBH	Sandakan	Sandakan, sebahagian Kota Kinabatangan	531,789	(A)
15	Gunung Andrassy	SBH	SBH	Tawau	Tawau, Pulau Sebatik	356,218	(A)

NO.	SITE NAME	STATE	REGION	LOCATION	COVERAGE AREA	AFFECTED POPULATION	CATEGORY (REFER TO TABLE B)
16	Bukit Tampin	NS	SOUTH	Tampin	Melaka, Port Dickson, Rembau, Tampin, Rantau, Pedas, Linggi, Senawang, Sebahagian Sepang and Sungai Pelek	942,745	(A)
17	Seberang Jaya	PNG	NORTH	Seberang Jaya	Georgetown, Jelutong, Gelugor, Batu Maung, Bayan Lepas, Seberang Jaya, Butterworth, Bukit Mertajam, Simpang Ampat, Kepala Batas, Kulim, Seberang Perai	901,445	(A)
18	Pelangi	JHR	SOUTH	Johor Bahru	Johor Bahru, Pasir Gudang	874,995	(A)

NO.	SITE NAME	STATE	REGION	LOCATION	COVERAGE AREA	AFFECTED POPULATION	CATEGORY (REFER TO TABLE B)
19	Telipot	KEL	EAST	Kota Bharu	Bachok, Kota Bharu, Pasir Mas, Tumpat, Ketereh	826,676	(A)
20	Bukit Larut	PRK	NORTH	Taiping	Bandar Baharu, Serdang, Selama, Parit Buntar, Bagan Serai, Semanggol, Taiping, Kuala Sepetang, Padang Rengas, Terong, Bandar Baharu, Kerian, Kuala Kangsar, Sungai Siput, Beruas, Pantai Remis	671,767	(A)
21	Skudai	JHR	SOUTH	Johor Bahru	Skudai, Senai, Johor Bahru	486,346	(A)

NO.	SITE NAME	STATE	REGION	LOCATION	COVERAGE AREA	AFFECTED POPULATION	CATEGORY (REFER TO TABLE B)
22	Penang U4	PNG	NORTH	George Town	Tanjung Tokong, Georgetown, Jelutong, Air Itam, Gelugor, Batu Maung	348,878	(A)
23	Bukit Lambir	SWK	SWK	Miri	Miri, Marudi, Sibuti, Bekenu, Lutong, Kuala Baram, Beluru	282,353	(A)
24	Bukit Nyabau	SWK	SWK	Bintulu	Bintulu, Sebauh, Tatau, Tanjung Kidurong Similajau	201,964	(A)

NO.	SITE NAME	STATE	REGION	LOCATION	COVERAGE AREA	AFFECTED POPULATION	CATEGORY (REFER TO TABLE B)
25	Bukit Bauk	TRG	EAST	Dungun	Dungun, Paka, Bukit Besi, Al Muktafi Billah Shah, Ketengah Jaya, Kerteh	153,123	(A)
26	Bukit Temuduk	SWK	SWK	Sri Aman	Betong, Spaoh, Sri Aman, Sebuyau, Lingga, Engkilili	108,874	(A)
27	Gunung Silam	SBH	SBH	Lahad Datu	Kunak, Lahad Datu	373,007	(A)

NO.	SITE NAME	STATE	REGION	LOCATION	COVERAGE AREA	AFFECTED POPULATION	CATEGORY (REFER TO TABLE B)
28	Bukit Lima	SWK	SWK	Sibu	Sibu, Bintangor	286,774	(A)
29	Bukit Fraser	PHG	EAST	Bukit Fraser	Raub, Benta, Damak, Sungai Koyan, sebahagian Kuala Lipis, sebahagian Jerantut, Kuala Kubu Baru, Batang Kali, Bestari Jaya, Bukit Beruntung, Serendah	436,077	(A)
30	Bukit Kelapa	SBH	SBH	Kudat	Kota Marudu, Kudat, Pitas	131,156	(A)

NO.	SITE NAME	STATE	REGION	LOCATION	COVERAGE AREA	AFFECTED POPULATION	CATEGORY (REFER TO TABLE B)
31	Bukit Tinggi	JHR	SOUTH	Mersing	Mersing, Endau, Kahang, Kluang, Jemaluang	138,691	(A)
32	Telapak Burok	NS	SOUTH	Jelebu	Sepang, Kuala Klawang, Port Dickson, Rembau, Kuala Pilah, Seremban, Jempol, Bera, Simpang Pertang	403,340	(A)
33	TM Exchange Kangar	PER	NORTH	Kangar	Padang Besar, Kangar, Arau, Changlun, Ayer Hitam, Simpang Ampat, Kodiang	257,363	(A)

NO.	SITE NAME	STATE	REGION	LOCATION	COVERAGE AREA	AFFECTED POPULATION	CATEGORY (REFER TO TABLE B)
34	Bukit Bintang	TRG	EAST	Jerteh	Pasir Putih, Cherang Ruku, Besut, Setiu, Jerteh, Permaisuri	232,648	(B)
35	Bukit Sembilan	PHG	EAST	Muadzam Shah	Bukit Ibam, Muadzam Shah, Chini, Bandar Tun Razak	74,294	(B)
36	Bukit Singgalang	SWK	SWK	Selangau	Mukah, Oya, Dalat, Stapang, Selangau	71,886	(B)

NO.	SITE NAME	STATE	REGION	LOCATION	COVERAGE AREA	AFFECTED POPULATION	CATEGORY (REFER TO TABLE B)
37	TM Sarikei	SWK	SWK	Sarikei	Bintangor, Meradong, Sarikei, Tanjung Manis, Selalang, Sebangkoi	72,526	(B)
38	Bukit Senggora	PHG	EAST	Maran	Maran, Chenor, sebahagian Jengka	74,530	(B)
39	Gunung Ancharang	SWK	SWK	Miri	Niah	38,403	(B)

NO.	SITE NAME	STATE	REGION	LOCATION	COVERAGE AREA	AFFECTED POPULATION	CATEGORY (REFER TO TABLE B)
40	Bukit Kemuning	TRG	EAST	Kemaman	Kemaman, Kijal, Kemasik, Ibok, Cherating	180,967	(B)
41	Gunung Raya	KDH	NORTH	Langkawi	Langkawi, Kuala Perlis, Simpang Ampat	89,353	(B)
42	Saratok	SWK	SWK	Saratok	Kabong, Kalaka, Pusa, Roban, Saratok, Saribas	49,378	(B)

NO.	SITE NAME	STATE	REGION	LOCATION	COVERAGE AREA	AFFECTED POPULATION	CATEGORY (REFER TO TABLE B)
43	Bukit Kapit	SWK	SWK	Kapit	Kapit	28,673	(B)
44	Kuala Rompin	PHG	EAST	Kuala Rompin	Kuala Rompin	27,996	(B)
45	Bukit Mas	SWK	SWK	Limbang	Limbang, Sundar, Lawas	51,847	(C)

NO.	SITE NAME	STATE	REGION	LOCATION	COVERAGE AREA	AFFECTED POPULATION	CATEGORY (REFER TO TABLE B)
46	Pulau Pangkor	PRK	NORTH	Pulau Pangkor	Pulau Pangkor, Teluk Batik, Lumut,Sitiawan	32,437	(C)
47	Belaga	SWK	SWK	Belaga	Belaga	3,387	(C)

Disclaimer: The information is generated using software tools based on 2020 Census Data.

^{*}Note: The number of the DTH connected user shall be updated every January and July of each year.